Working Paper

Number 11/02 Mexico, January 2011

Economic Analysis

Forthcoming, *Review of International Economics*

Effects of a Free Trade Agreement on the Exchange Rate Pass-Through to Import Prices

Arnoldo López-Marmolejo

Forthcoming, Review of International Economics

Effects of a Free Trade Agreement on the Exchange Rate Pass-Through to Import Prices

Arnoldo López-Marmolejo *

January 2011

This paper investigates the effect of trade liberalization on the exchange rate pass-through (ERPT) to import prices. To do so, it employs an empirical estimation of the effects of NAFTA on the Mexican ERPT, and uses a Ricardian general equilibrium model. The model identifies two channels that explain how the trade liberalization alters the ERPT. The first channel is the direct relationship between the tariffs and the pass-through by good. The second channel is the effect that tariffs have on the composition of imports, altering indirectly the aggregate pass-through.

Keywords: Ricardian model, exchange rate pass-through, NAFTA.

JELS F31 and F41.

^{*} I thank Hugo Rodríguez, Timothy Kehoe, Víctor Ríos-Rull, Juan Carlos Conesa, and an anonymous referee for helpful comments.

1. Introduction

This paper explores the effects of trade liberalization on the "exchange rate pass-through" (ERPT) to import prices. The ERPT is the responsiveness of prices to exchange rate movements. It can be measured in terms of elasticities: the percentage change in the domestic prices of imported goods resulting from a 1% change in the nominal exchange rate between the importing and exporting countries. Trade liberalization is seen in this paper as a two-country tariffs elimination. The contributions of this paper are as follows. First, based on a model, it proposes two channels through which tariffs elimination can alter the ERPT. The first channel is a decrease in the ERPT levels by product. The second channel is a change in the composition of the import bundle, altering the aggregate ERPT. Second, in terms of economic theory, it develops a model that mixes trade theory and new open economy models. Third, it incorporates a case analysis to show that a trade agreement can have an important effect on the ERPT to import prices.

I will begin with a brief review of the literature on ERPT in order to situate the analysis on this literature. The effects of the ERPT on import prices have been analyzed as both a macro phenomenon and a micro phenomenon. In the 1990s, microeconomic studies emphasized industrial organization, price discrimination, and geographical segmentation across product markets as causes for the ERPT. For a survey of this literature see Goldberg and Knetter (1997). More recently, the ERPT has become a central issue in the debates over the election of exchange rate regimes and appropriate monetary policies. One recent hypothesis suggests that the intensity of the ERPT depends on the country's relative monetary stability,¹ that is, once inflation is stabilized, the ERPT to prices becomes weaker since the firm's power to determine the prices is weaker. Choudhri and Hakura (2006) find evidence to support this hypothesis for a sample of 71 countries. Goldfajn and Werlang (2000) and Baqueiro, Díaz de León and Torres (2004) also support this hypothesis using cross-section analysis and timeseries analysis, respectively. Furthermore, they show that exchange rate fluctuations are also an important determinant of the ERPT. Finally, another recent study on the determinants of the ERPT is Campa and Goldberg (2005). For a sample of 25 OECD countries, they find that the most important determinants are microeconomic. The countries with a relatively large share of raw materials and energy products in their imports are the ones with higher levels of ERPT. In this paper, I go a step backwards. I suggest that the tariffs structure alters the composition of a country's imports, causing some effects on the aggregate pass-through level. However, I also explore how tariffs elimination affects the ERPT level by product and how this individual effect interacts with the new import bundle to determine the aggregated level of pass-through. Answering these questions is particularly important given the widely extended trade liberalization we see these days.

Moreover, a model to explain the effects of a free trade agreement (FTA) is needed as the relationship between trade liberalization and pass-through is not clear. On one hand, some economists claim that trade liberalization diminishes the pass-through level because it increases competition, thereby reducing the power of firms to move prices. However, on the other hand, empirical evidence (see De Stefano, 2004, and Kardasz and Stollery, 2005) shows that the higher the market power of the firms, the lower the pass-through of exchange rate movements to the final prices. The last result would imply that if the trade liberalization affects the ERPT through an increase in competition, trade liberalization would increase the pass-through. In addition, when analyzing the effects of trade barriers on the pass-through, different studies suggest opposite effects; see, for example, Kardasz and Stollery (2005) and Barhoumi (2006). Therefore, explaining how trade liberalization affects the ERPT is still an open question.

¹ See Taylor (2000) and Devereux and Engel (2001).

A Ricardian two-country model in which there are two types of tradable goods with different pass-through levels is used. The ERPT is modeled on Corsetti and Dedola (2005). That is, an incomplete pass-through arises as a consequence of the need for local distribution services and wage rigidity. The inclusion of the distribution sector is motivated by the importance of this sector in consumer price formation. The empirical evidence for the US (see Burstein, Neves and Rebelo, 2003) shows that the distribution costs represent more than 50% of the retail price in all agricultural products and that the total distribution margin is around 42%. The next section discusses the model, although, some clarifications are needed. For simplicity, I focus on an FTA between two symmetric open economies. Each economy has two tradable sectors: one sector producing high-differentiation goods and another producing low-differentiation goods. Following Campa and Goldberg (2005) and Oliveira, Scarpetta and Pilat (1996), the products that are considered as high-differentiation goods are the manufacturing and high-tech products, and the products that are considered as low-differentiation goods are agricultural products and raw materials. The firms producing tradable goods face monopolistic competition. By assumption, the firms in each country choose a price for each market, domestic and foreign, taking into account the market demand but not the foreign competition. This restriction for the firms with regard to knowing the foreign competition does not seem to be highly unrealistic as the firms are located in different countries. As such, the producers of tradable goods in the other country can enter the market and make positive profits while some domestic producers cannot. This can be considered as a non-Nash behavior of the firms owing to the information restriction on foreign competitors. However, due to the Ricardian production functions, some firms in each country always have a comparative advantage; consequently, there will be imported goods no matter what the rest of the firms do and therefore the results of the paper are not altered due to this assumption.

The model works is as follows. Due to the FTA, both countries import more varieties of lowand high-differentiation goods from the other country. The high-differentiation products have the lowest ERPT to import prices.² When the tariffs on the low-differentiation goods sector are higher as compared to those on the high-differentiation goods sector, the number of varieties of high-differentiation goods that are imported increases more than the number of varieties of low-differentiation goods that are imported, thereby decreasing the aggregate ERPT. In addition, the reduction in tariffs decreases the ERPT in each imported good. This occurs because in the model, tariffs have a positive effect on the price elasticity of demand faced by tradable firms and the elasticity is associated with the ERPT. The intuition is that ad-valorem tariffs increase the proportion of the final price influenced by the foreign firm in relation to the share of the price resulting from distribution costs. As the foreign firms have more influence on the final price, they pass-through a higher proportion of the exchange rate movements into import prices since the firm compensates the change in its profits resulting from exchange rate fluctuations with the movements in prices. The simulations performed show that tariffs elimination reduces the pass-through level by good and by sector.

In this model, the demand curve is different across sectors due to the differences in tariffs and market power, and the ERPT varies across sectors because the firms take into account these conditions when they move the price in response to exchange rate movements. In contrast, there are models that rely on the adjustments along the demand curve to model ERPT. For example, in the model of Auer and Chaney (2009), when firms adjust their prices, they move along the demand curve and face different demand elasticities due to the exit of low-quality firms. Exporters adjust their prices in response to the level of competition (number of firms in the market) and cost changes brought about by exchange rate fluctuations.

The rest of the paper is organized as follows: section two presents the case analysis, section three contains the model, section four explains the calibration procedure, and section five gives the numerical results. Finally, section six concludes with a brief summary of the results and the ideas contained in this paper.

² See Campa and Goldberg (2005) for the empirical evidence.

2. Empirical Evidence

The purpose of this section is to use a case analysis to explore empirically the effect of an FTA on the ERPT to import prices. To my knowledge, there is no paper that has attempted to use a case to analyze this issue. The case chosen is the North America Free Trade Agreement (NAFTA) and its effects on the ERPT in Mexico. The case of Mexico is chosen because its high integration with USA provides the following advantages in terms of ERPT measurement. It permits the measurement of the ERPT using an aggregate import price index and yields the possibility of focusing on the transmission of a shock from a single currency (US dollar) movement.

After 1986, Mexico chose the liberalized economy approach. On August 17, 1986, Mexico joined the General Agreement on Trade and Tariffs (GATT) and on January 1, 1994, NAFTA came into effect.

Thanks to these agreements, the trade relationship between Mexico and the USA became more and more important. Graphic 1 shows how the trade of Mexico with the USA started growing after 1987.





Source: own calculations

This strong trade relationship ended with the USA becoming the most important trading partner of Mexico, and Mexico becoming the third-most important trading partner of the USA after Canada and China, when trade is measured as the sum of the exports and imports.

With regard to the total Mexican imports, around 70% of the Mexican imports are sourced from the USA. This hard-to-get characteristic implies that the import price index basically represents the pricing decisions of the USA firms for the Mexican market. As such, this index is used to analyze the ERPT behavior when the bilateral trade had been liberalized. Diversified economies' import price indexes would not be capable of capturing the effect of bilateral trade liberalization. Note that an essential characteristic of the import prices variable is its domestic currency denomination, which makes it possible to obtain the final effect on the domestic prices.

The model described in the next section and papers such as Campa and Goldberg (2005) shows that the prices of imported goods depend on their respective production costs, prices in the distribution sector, exchange rate, and market power. The data used in the estimation are as follows. As a dependent variable, I employ an import price index calculated by the

OECD from the prices denominated in the local currency. The exchange rate is denominated in Mexican pesos by US dollars. As a marginal cost, I use the USA payroll index. Due to the lack of a retail sector price index in Mexico, the proxy variable for the price of the distribution services is the Mexican salary index. The producer prices were not chosen as the proxy because this index contains imported goods. The sources of the data appear in appendix 1.

The behavior of the series is analyzed in order to define the estimation model. An analysis of unit root is performed to test whether the series follows a unit root or a stationary trajectory. The unit root analysis consists of three tests, the augmented Dickey-Fuller (ADF) test, Perron test, and Zivot-Andrews test. The ADF is the traditional unit root test. The Perron and the Zivot-Andrews tests take into account the presence of structural breaks. Perron's (1989) procedure is characterized by a single exogenous break, while Zivot and Andrews (1992) selects the break date endogenously. The break date is given by the point where the t-statistic from the ADF test of unit root is at the minimum (most negative). The next table shows that the ADF test rejects the null hypothesis of the unit root process of the following variables: import prices, US payrolls, and Mexican salaries. However, the Perron and Zivot-Andrews tests do not reject the null hypothesis of the unit root process for the import prices, exchange rate, and US payrolls. For the Mexican salaries, the tests give opposite results.

Chart 1

Test for Unit Root of Residuals

Unit root process. Ho: variable contains a unit root				
Variables	Augmented Dickey-Fuller test	Perron test	Zivot-Andrews test	
Import prices	3.657	1.197	3.755	
	(1.950)	(3.467)	(5.080)	
Exchange rate	0.088	1.696	3.359	
	(1.950)	(3.467)	(5.080)	
US payrolls	13.768	1.549	2.738	
	(1.950)	(3.467)	(5.080)	
Mexican salaries	4.753	1.508	7.056	
	(1.950)	(3.467)	(5.080)	

Numbers in parentheses represent the critical value

Variables in logs

Graphic 2

Source: own calculations

The Zivot-Andrews test also identifies the break date. The break in the import prices and the break in the exchange rate coincide with the signing of the NAFTA and the Mexican crisis in 1994-1995. We can consider these breaks as shocks that had persistent effects on the longrun level of the variables.



Break date (t-statistic from the ADF test of unit root)

Source: own calculations

By definition, the difference in a variable that follows a unit root process is stationary. As such, in such a case, an estimation in first differences is chosen.

Thus, the ERPT is calculated as follows:

 $\Delta imp_{j,t} = \beta_0 + \beta_1 \Delta imp_{j,t-1} + \beta_2 \Delta e_t + \beta_3 \Delta c_{j,t} + \beta_4 \Delta pn_{k,t} + \nu_{j,t}, \, k \neq j,$

where Δ represents the first differences, *imp* denotes the import prices in country *i* denominated in the domestic currency, e denotes the Peso/USD exchange rate, c, is the marginal cost in country j, pn is an indicator of the price of the distribution services in the trade partner k, and v_i is the error term, and all variables are log-transformed. The short-term ERPT is given by the estimated coefficient β_{γ} .

The estimation period is given by the guarterly data from the first guarter of 1987 to the second quarter of 2003. The reasons for choosing the first quarter of 1987 are that the 1980s Mexican crisis came to its end in this period and that the trade liberalization had just begun in this year. The estimation period ends in 2003 because the import prices calculated by the OECD were discontinued after that year. I use this index because it is the only one available that is denominated in the domestic currency and this characteristic permits the capturing of the exchange rate transmission to the final price. The OECD dataset contains three import price indexes for Mexico, total goods, manufacturing goods, and other goods. Here, manufacturing goods comprise machinery, transport equipment, electronics, and intermediate manufacturing goods, and other goods comprise food products, beverages, textiles, and energy products.

The estimation equation in this analysis is very similar to those used by Campa and Goldberg (2005) for OECD countries, Otani, Shiratsuka and Shirota (2003) for Japan and Bugamelli and Tedeschi (2008) for Europe.

A test to analyze multicollinearity is performed. The Variance Inflation Factor (VIF) shows no evidence of multicollinearity among the independent variables (see chart 2). Then, the control variables do not seem to have a strong linear relationship among them.

Variable	VIF	1/VIF
D.Total goods import prices _{t-1}	1.32	0.76
D.Exchange rate,	1.23	0.81
D.Mexico salaries,	1.14	0.88
D.US Payrolls,	1.05	0.95

Chart 2 Multicollinearity test

The estimation results are as follows:

Chart 3 Estimation results

	Total goods import prices _t	Manufacturing goods import prices,	Other goods import prices,
D.Exchange rate,	0.984	0.997	0.913
- ((24.50)***	(29.31)***	(1.87)*
D.US payrolls,	0.346	0.672	-4.207
·	(1.14)	(2.18)**	(1.38)
D.Mexican salaries,	0.003	-0.021	0.46
·	(0.07)	(0.54)	(2.04)**
D.Total goods import prices	0.013		
e 1	(0.43)		
D.Manufacturing goods import prices		0.017	
		(0.72)	
D.Other goods import prices _{t-1}			-0.476
			(1.80)*
Constant	0.001	-0.002	0.051
	(0.14)	(0.62)	(1.43)
Observations	66	66	66
R-squared	0.94	0.95	0.35

Robust t statistics in parentheses. Variables in logs. D denotes in differences. * significant at 10%; ** significant at 5%; *** significant at 1%.

Manufacturing goods refer to machinery, transport equipment, electronics, and manufacturing intermediate goods.

Other goods refer to food products, beverages, textiles, and energy products. Source: own calculations

The ERPT is significant for each series of import prices. In the case of the prices of manufacturing imports, the US Payroll index has a significant and relevant positive sign. For the other goods, the Mexican costs are the ones with a positive and significant coefficient. This implies that the domestic costs are more important for the determination of the prices of manufacturing imports whereas the foreign distribution costs are more important for the other goods price formation. It seems that the aggregation of manufacturing and other goods in the total imports causes the effect of the costs on the aggregate import prices to cancel out.

In the case of serial correlation, the estimation results might be subject to potential problems of biased and inconsistent estimators. However, there is no evidence of serial correlation in the estimation of the ERPT for total goods and for manufacturing goods whereas for other goods, there is evidence of only week serial correlation (autocorrelation of the residuals around 0.34). An alternative to eliminate the problems resulting from serial correlation is to use instrumental variables for the suspected endogenous variables.³ Regardless of the weak evidence of serial autocorrelation, the ERPT was also estimated using a two-stage least squares (2SLS)⁴ method yielding asymptotically efficient estimates of the coefficients, including the first and second lags of the suspected endogenous variables as instruments in the regression.⁵ The results were the same as in the previous estimation concluding that the serial correlation is not biasing the results.

To analyze the changes in the ERPT over time, I employ the rolling regression method. First, the model estimates the pass-through using a 24 observation subsample. The model then extends the sample to an additional period and later performs the estimation. The process is repeated till the end of the sample. The subsample estimation incorporates the data from the first quarter of 1987 to the fourth quarter of 1992 (24 observations).

Graphic 3 shows that the ERPT seems to be decreasing over time. In particular, we observe two different and persistent levels of pass-through before and after NAFTA. The pass-through level before NAFTA is close to 1.10 and it is 0.98 after NAFTA.





Source: own calculations

⁵This is particularly true in our case given the small size of our sample. In the case of large samples, the generalized method of moments (GMM), developed by Hansen (1982) provides efficient estimators.

³ From among the works on the ERPT, Mihailov (2009) applies this technique. The recent studies have also used vectorautoregressions to analyze this issue (see Choudhri, Faruquee and Hakura, 2005, Coricelli, Jazbec and Masten, 2006, McCarthy, 2007, Ito and Sato, 2008). Among the most common critiques to this method is the lack of a theoretical model to define the estimation equations, and in particular, the lags used in the estimation.

⁴This estimator is also known as the Anderson and Hsiao (1981) estimator.

Trade liberalization includes the elimination of both tariff and non-tariff barriers. It is important to note that the trade liberalization of different products came at different points in time due to legal agreements and organizational restrictions, a reason that can explain why the pass-through decreases sharply over time in the first years of NAFTA. Graphic 3 also shows a jump in the ERPT in the first quarter of 1995, when the Mexican crisis begins. However, in spite of the prevailing turbulence during 1995 and 1996 (for example, an inflation rate of 52% in 1995 and 28% in 1996), the pass-through has a stable pattern around 0.98.

The rolling regression method allows us to calculate an ERPT coefficient for each period of time. Consequently, I can try to explain this coefficient. In Mexico and the USA, the variables that can be used to explain the ERPT is inflation, the trade over GDP as a measure of protectionism in a country (tariff and non-tariff trade barriers are not available over time and tariffs are altered only after long periods), and the share of manufacturing imports in total imports.

The Durbin's test for serial autocorrelation shows that the residuals follow an AR(1) process. A feasible generalized least squares (FGLS) estimation is done to take into account this process.

Trade over GDP,	-0.662
	(7.97)***
Mexico inflation,	0.002
	(3.23)***
US inflation,	-0.007
	(0.02)
Share of manufactures in total imports,	0.174
	(1.12)
Constant	1.086
	(84.40)***
Observations	43
R–squared	0.76

Chart 4 Quarterly Exchange Rate PassThrough, : FGLS Estimation

Absolute value of z statistics in parentheses.

* significant at 10%; ** significant at 5%; *** significant at 1%.

Source: own calculations

The estimation shows that the level of trade liberalization, measured as trade over GDP, has a statistically significant negative effect on the ERPT in Mexico. The level of inflation is significant; however, its coefficient is very close to zero. The US inflation and the share of manufactures in total imports are not significant. The VIF test (see chart 5) does not show evidence of multicollinearity among variables used in this regression.

Multicollinearity test: ERPT regression Variable VIF 1/VIF 1.48 0.68 Mexico inflation, Share of manufactures in total imports, 1.48 0.68 Trade over GDP 1.08 0.92 US inflation, 1.01 0.99

Source: own calculations

Chart 5

A change in the import composition should alter the ERPT since different goods have different pass-through levels. Campa and Goldberg (2005) have shown that this seems to be the case in many OECD countries, wherein the ERPT shows a decreased owing to the higher proportion of manufacturing imports. However, in the case of Mexico, manufactures represent more than 85% of the total imports since 1987 and this measure has been highly persistent, in particular in the last decade. This is possibly the reason for the no significance of the share of manufacturing in the total imports coefficient. Further, it must be noted that the concentration in the manufacturing sector does not necessarily imply a reduction of the ERPT; the ERPT depends on the pass-through level in this sector. The estimation results in chart 3 show that in Mexico, the manufactured goods have a high level of ERPT (0.9) as compared to in other countries (see Campa and Goldberg, 2005).

An alternative index of import prices is calculated by the Central Bank of Mexico. In converse to the index calculated by the OECD, this index is calculated from the prices denominated in US dollars, which makes it more difficult to perceive the transmission of the exchange rate shocks to the final price. In order to derive a contrast to our previous results, the ERPT is calculated using this index in spite of its limitations. The estimation is done following the same procedure described above and uses the same control variables. The estimation results using this monthly index show the same decreasing pattern in the ERPT after NAFTA, as found previously. The estimation also shows an additional reduction in the ERPT in the crisis period of 2009-2010, which may be associated with the demand factors. The ERPT is smaller because it is a monthly pass-through and the index is calculated from the prices in US dollars.

Graphic 4





Source: own calculations

3. Model

A two-country Ricardian model similar to the one described in Corsetti and Dedola (2005) is used. Each country has four sectors: two tradable sectors (one producing high-differentiation products and another producing low-differentiation products), a nontradable sector, and a distribution sector. The firms in each of the tradable sectors produce their goods in a number of varieties or brands; this implies that in each tradable sector the goods (varieties) are slightly different across them. By definition, the high-differentiation goods are more heterogeneous than low-differentiation goods. The firms producing tradable goods sell these goods in both the domestic market and the foreign market. In its own market, the domestic firm sells its good directly to the consumer. In the foreign market, the domestic firm sells the product to a foreign distribution firm. This distribution firm buys the tradable good. The nontradable goods are used to make foreign tradable goods available to domestic consumers. One can consider this nontradable good as the service required to distribute and sell a good in the foreign country with the same conditions as in the home country.

Separating the tradable firms from the distribution firms results in two different import prices, the producer price and the consumer price. The producer price is represented by the price that the distribution firm receives from the tradable firm. It does not include the taxes and the distribution costs. The consumer price is given by the final price and includes both the taxes and the distribution costs.

The decision to consider both the producer import price and the consumer import price deserves some discussion. I decided to do so in a way that allows me to calculate an import price consistent with the one calculated by the Bureau of Labor Statistics (BLS) and other international statistical offices. This will allow me to compare the results with the empirical evidence. The BLS reports the import prices on a free on board (f.o.b.) foreign port basis. The f.o.b. foreign port price is the price at the foreign port of export before insurance, freight, or duty are added. Although the f.o.b. foreign port price bases such as cost, insurance, freight (c.i.f.) for imports as long as the firms can provide consistent price series. However, both the f.o.b. foreign port price series and the c.i.f. USA port price series exclude duty as well as costs associated with domestic intermediaries (for example, wholesalers and retailers). All these factors may affect the final selling price. For purposes of deflating imports, however, the duties are excluded from the prices before the indexes are calculated. Then, the definition on the import prices.

The economy is represented by three types of agents in each country: households, firms, and a government. The households in each country own the firms in their respective country. The firms producing tradable goods are monopolistic suppliers of only one variety of goods. For simplicity, I assume that firms producing nontradable goods produce a homogeneous good and face perfect competition. The government transfers its revenues to the households in a lump-sum fashion.

3.1. Households

Both high-differentiation and low-differentiation goods sectors are represented by a continuum of individual differentiated goods, indexed by $h \in [0, 1]$ and $a \in [0, 1]$, where *h* represents a good in the high-differentiation goods sector and *a* represents a good in the low-differentiation goods sector. In addition, each country produces a nontradable good, denoted by *n*. Country 1 is populated by a continuum of households indexed by $i \in [0, 1]$. Similarly, country 2 is populated by differentiated households indexed by $j \in [0, 1]$.

The household *i* in country 1 consumes differentiated tradable goods. Let $C_{1,A,i}(i)$ and $C_{1,H,i}(i)$, respectively, represent the consumption of low-differentiation and high-differentiation tradable goods in country 1 by the household *i* in period *t*. These consumption indexes depend on the consumption of the low-differentiation good *a*, $c_{1,i}(a, i)$, and the consumption of the high-differentiation good *h*, $c_{1,i}(h, i)$, in the following manner:

$$\begin{split} C_{1,A,t}(i) &= \left(\int_0^1 c_{1,t}(a,i)^{\frac{\psi-1}{\psi}} da\right)^{\frac{\psi}{\psi-1}},\\ C_{1,H,t}(i) &= \left(\int_0^1 c_{1,t}(h,i)^{\frac{\theta-1}{\theta}} dh\right)^{\frac{\theta}{\theta-1}}. \end{split}$$

The full-consumption basket of household *i* is

$$C_{1,t}(i) = C_{1,H,t}^{\alpha}(i)C_{1,A,t}^{1-\alpha}(i).$$

The prices in country 1 of goods *a* and *h* are denoted by $p_{1,i}(a)$ and $p_{1,i}(h)$, respectively. The price indexes for the previous consumption indexes are given by

$$P_{1,A,t} = \left(\int_0^1 p_{1,t}(a)^{1-\psi} da\right)^{\frac{1}{1-\psi}},$$
$$P_{1,H,t} = \left(\int_0^1 p_{1,t}(h)^{1-\theta} dh\right)^{\frac{1}{1-\theta}},$$
$$P_{1,t} = \frac{P_{1,H,t}^{\alpha} P_{1,A,t}^{1-\alpha}}{\alpha^{\alpha} (1-\alpha)^{1-\alpha}}.$$

Taking into account the previous notation, the representative household i in country 1 maximizes the utility function

$$U_{1,t}(i) = Eo \sum_{t=0}^{\infty} \beta^t \left[\log C_{1,t}(i) + x \log \left(\frac{M_{1,t}(i)}{P_{1,t}} \right) - k l_{1,t}(i) \right], \tag{1}$$

which depends positively on the consumption basket $C_{1,i}(i)$ and the real balances $\frac{M_{1,t}(i)}{P_{1,t}}$, and negatively on the labor $l_{1,i}(i)$. The parameter $\beta \in (0, 1)$ is the discount factor.

Household *i* in country 1 faces the following budget constraint:

$$M_{1,t}(i) + B_{1,t+1}^{1}(i) + \mathcal{E}_{t}B_{1,t+1}^{2}(i) + P_{1,t}(i)C_{1,t}(i) \leq M_{1,t-1}(i) + w_{1,t}(i)l_{1,t}(i) + (1+r_{1,t})B_{1,t}^{1}(i) + (1+r_{2,t})\mathcal{E}_{t}B_{1,t}^{2}(i) + \int_{0}^{1} \pi_{1,t}(h,i)dh + \int_{0}^{1} \pi_{1,t}(a,i)da,$$
(2)

where $w_{1,i}(i)$ is the wage associated with the time dedicated to the work by household *i*, $T_{1,i}(i)$ denotes the transfers made by the government, and $\int \pi_{1,i}(h, i)dh$ and $\int \pi_{1,i}(a, i)da$ are the dividends from the equities of all domestic firms *h* and *a*, respectively, in the economy. The domestic currency is represented by $M_{1,i}(i)$. The nominal exchange rate \mathcal{E}_i , defined as the home-currency price of the foreign currency, is used to denominate the prices in the foreign currency. Let $B_{1,i}^1(i)$ and $B_{1,i}^2(i)$ denote the two international bonds denominated in the domestic currency and the foreign currency, respectively, owned by household *i*. The interest rates paid by these bonds are given by $r_{1,i}$ and $r_{2,i}$. The total bonds owned by household *i* in country 1 are then given by $B_{1,i}^1(i) + B_{1,i}^2(i) = B_{1,i}(i)$. The same notation applies for the representative agent in country 2. Define the total income of the household in country 1 at time t as

$$\begin{split} I_{1,t}(i) &= M_{1,t-1}(i) - M_{1,t}(i) + w_{1,t}(i)l_{1,t}(i) + T_{1,t}(i) - B_{1,t+1}^1(i) - \mathcal{E}_t B_{1,t+1}^2(i) \\ &+ (1+r_{1,t})B_{1,t}^1(i) + (1+r_{2,t})\mathcal{E}_t B_{1,t}^2(i) + \int_0^1 \pi_{1,t}(h,i)dh + \int_0^1 \pi_{1,t}(a,i)da \\ \end{split}$$

A similar condition applies for the households in country 2. Given the utility function (1), the demands for products a and h in household i are given as

$$c_{1,t}(h,i) = \left(\frac{\alpha I_{1,t}(i)}{p_{1,t}(h)}\right)^{\theta} \frac{1}{C_{1,H,t}^{\theta-1}}, c_{1,t}(a,i) = \left(\frac{(1-\alpha)I_{1,t}(i)}{p_{1,t}(a)}\right)^{\psi} \frac{1}{C_{1,A,t}^{\psi-1}}.$$
(3)

As such, ψ and θ represent the elasticities of demand with respect to price. It is assumed that $\psi > \theta > 1$ since low-differentiation goods are by definition more homogeneous than high-differentiation goods, and therefore face higher elasticity levels. In this type of models, ψ and θ also represent the market power of the firms. The closer ψ comes to ∞ , the more competitive the sector is.

For household *i*, the first-order conditions with respect to the bonds demand imply that

$$\frac{1}{C_{1,t}(i)} = \beta E_t \left(\frac{1+r_{1,t+1}}{P_{1,t+1}C_{1,t+1}(i)}\right) P_{1,t},\tag{4}$$

$$\frac{\mathcal{E}_t}{C_{1,t}(i)} = \beta E_t \left(\frac{\mathcal{E}_t(1+r_{2,t+1})}{P_{1,t+1}C_{1,t+1}(i)} \right) P_{1,t}.$$
(5)

The first-order condition with respect to the money demand is given by

$$M_{1,t}(i) = x \frac{(1+r_{1,t+1})}{r_{1,t+1}} P_{1,t} C_{1,t}(i).$$
(6)

Similarly, for household j in country 2, the first-order conditions are given by

$$\frac{1}{C_{2,t}(j)} = \beta E_t \left(\frac{1 + r_{2,t+1}}{P_{2,t+1}C_{2,t+1}(j)} \right) P_{2,t},\tag{7}$$

$$\frac{1}{C_{2,t}(j)} = \beta E_t \left(\frac{1 + r_{1,t+1}}{P_{2,t+1}C_{2,t+1}(j)\mathcal{E}_{t+1}} \right) P_{2,t}\mathcal{E}_t,\tag{8}$$

$$M_{2,t}(j) = x \frac{(1+r_{2,t+1})}{r_{2,t+1}} P_{2,t} C_{2,t}(j).$$
(9)

3.2. Firms

It is worth remembering that this economy has four different sectors: one producing highdifferentiation products, one producing low-differentiation products, one nontradable goods sector, and one distribution sector. While the firms in the distribution sector buy tradable goods and use the nontradable goods as an input, the firms in the tradable and nontradable sectors produce using only labor. Let $y_{i}^{u}(h)$, $y_{i}^{u}(a)$, and $y_{i}^{u}(n)$ denote the total outputs of the goods *h*, *a*, and *n*, respectively, produced by country u = 1, 2. The total output can be separated by its country of consumption. For example, $y_{1,i}^{u}(h)$ and $y_{2,i}^{u}(h)$ represent the production of good *h* in country *u* that is sold in country 1 and in country 2, respectively. Then, $y_{i}^{u}(h) = y_{1,i}^{u}(h) + y_{2,i}^{u}(h)$. This notation will be useful when solving the firms' problems. A similar notation applies for the low-differentiation sector. The demands for the labor inputs of type *i* by a firm producing goods

BBVA Research

h, *a*, and *n* are given by $l_{u,t}(h, i)$, $l_{u,t}(a, i)$, and $l_{u,t}(n, i)$, respectively. The production functions of the firms in each sector are given by

$$y_t^u(h) = y_{1,t}^u(h) + y_{2,t}^u(h) = \frac{1}{b_u(h)} \left(\int_0^1 l_{u,t}^{\frac{\varphi-1}{\varphi}}(h,i) di \right)^{\frac{\varphi}{\varphi-1}}, \ u = 1, 2,$$
(10)

where

 $b_1(h) = \gamma + \lambda h,$

$$b_{2}(h) = \gamma + \lambda(1-h);$$

$$y_{t}^{u}(a) = y_{1,t}^{u}(a) + y_{2,t}^{u}(a) = \frac{1}{b_{u}(a)} \left(\int_{0}^{1} l_{u,t}^{\frac{\varphi-1}{\varphi}}(a,i) di \right)^{\frac{\varphi}{\varphi-1}}, u = 1, 2,$$
(11)

where

$$b_1(a) = \gamma + \lambda a,$$

$$b_2(a) = \gamma + \lambda (1 - a),$$

$$y_t^u(n) = \left(\int_0^1 l_{u,t}^{\frac{\varphi - 1}{\varphi}}(n, i) di\right)^{\frac{\varphi}{\varphi - 1}}.$$
(12)

Note that φ represents the elasticity of substitution among the labor inputs, and $\frac{1}{b_u(q)}$, q = h, a; u = 1, 2, denote the productivity parameters that are variety specific. These allow each country to be more efficient in producing some varieties of the goods than the other country. This occurs because the productivity $b_u(q)$, q = h, a; u = 1, 2, depends on the good q and moves in the opposite directions in each country depending on the variety. Thus, these productivity parameters also imply that the varieties have been lined in the interval [0,1] depending on the differences in their productivities across countries. The varieties with productivities such that their production costs are lower in country 1 than in country 2 are located on the left side of the interval [0,1], whereas the varieties with productivities such that their production functions for the tradable sectors are referred to as the Ricardian production functions.

This mechanism offers a comparative advantage different from the classical Ricardian model by Dornbusch et al. (1977). While in the mechanism of this model, producing a good requires the same amount of labor across countries but there are differences in the productivity of each country due to variety-specific differences in technology, in the classical mechanism, the amount of labor required to produce one unit of the good is different across countries. Nevertheless, both mechanisms result in lower unit costs for certain varieties in each country.

3.2.1. Wage rigidity

Before explaining the firm's problem by sector, I analyze the wage setting. Nominal rigidities are allowed by assuming that wages in period *t* are predetermined using the contracts signed at time t - 1. When setting the wage w(i), each domestic worker acts as a monopolistic supplier of his/her productive input. That is, the worker takes into account the firm's demand in the utility maximization problem.

BBVA Research

The demand for labor *i* by domestic firm *h* in country 1 is obtained by minimizing the expenditure on labor in firm *h* as follows

$$l_{1,t}(h,i) = \left(\frac{w_{1,t}(i)}{W_{1,t}}\right)^{-\varphi} l_{1,H,t},$$

where $W_{1,t} = \left(\int_{0}^{1} w_{1,t}^{1-\varphi}(i)di\right)^{\frac{1}{1-\varphi}}$ is the country 1 wage index and $l_{1,H,t} = \left(\int_{0}^{1} l_{1,t}^{\frac{\varphi-1}{\varphi}}(h,i)di\right)^{\frac{\varphi}{\varphi-1}}$ is the total labor demand by firm *h*. Since φ is the same across sectors, the total demand for labor *i* is given by

$$l_{1,t}(i) = \left(\frac{w_{1,t}(i)}{W_{1,t}}\right)^{-\varphi} \left(\int_0^1 l_{1,H,t} dh + \int_0^1 l_{1,A,t} da + l_{1,N,t}\right),$$
(13)

where $l_{1,A,t} = \left(\int_0^1 l_{1,t}^{\frac{\varphi-1}{\varphi}}(a,i)di\right)^{\frac{\varphi}{\varphi-1}}$ and $l_{1,N,t} = \left(\int_0^1 l_{1,t}^{\frac{\varphi-1}{\varphi}}(n,i)di\right)^{\frac{\varphi}{\varphi-1}}$ are the total labor demands by firms a and n, respectively.

The workers are assumed to be monopolistic suppliers of a particular type of labor. Thus, they take into account the above labor demand in their utility maximization problem. The first-order condition for the optimal wage contract then becomes

$$w_{1,t}(i) = \frac{\varphi}{\varphi - 1} E_{t-1} \left\{ k P_{1,t} C_{1,t}(i) \right\}.$$
 (14)

The expected value at period t - 1 in the right-hand side of the above equation indicates that the wage contracts are set one period in advance.

Once the nominal wages are set, the workers are willing to meet any unanticipated changes in the firm's labor demand, given that the real wages do not fall below the marginal rate of substitution between leisure and consumption:

$$\frac{w_{1,t}(i)}{P_{1,t}C_{1,t}(i)} \ge k.$$
(15)

Therefore, the workers set nominal wages one period in advance, and supply the amount of labor that firms demand at the posted wage ex-post such that (15) is satisfied.⁶

3.2.2 Distribution sector

The distribution firm buys foreign tradable products, pays import taxes, and distributes them to the consumers using the nontradable good. For simplicity, I assume that these firms face perfect competition. As in Erceg and Levin (1995), Burstein, Neves and Rebelo (2003), and Corsetti and Dedola (2005), I assume that making available one unit of a traded good to the consumers requires δ units of a nontradable good.

Due to the distribution costs and the import taxes, there is a wedge between the producer import price and the consumer import price of the traded goods. Define $\tilde{p}(h)$ and $\tilde{p}(a)$ as the producer prices of goods *h* and *a*, respectively. Further, there is an extra cost for the foreign products sold in the home market, and vice versa, specifically, an ad-valorem tariff.

The origin and the destination of the products are important for the firms for two reasons. (i) In the presence of local distribution services in the foreign market, the tradable firms find it optimal to discriminate their prices across the markets. (ii) The distribution firm only sells the foreign goods by construction. Taking into account the origin of the product and the previous notation, equation (3) can be written as $c_{1,t}^u(h,i) = \left(\frac{\alpha I_1,c(i)}{p_{1,t}^v(h)}\right)^{\theta} \frac{1}{c_{1,t,n}^{1-1}}$, u = 1,2. This equation gives the demand in household *i* for good *h* that is produced in country *u*. Let $p_{-1}^u(h)$ be the price in country 1 of a

⁶ The parameter k indicates the magnitude of the disutility of labor and it is chosen to obtain labor magnitudes that correspond to the data. The calibration process of labor and the remaining variables allows reasonable relative magnitudes among the variables. Therefore, the assumption in equation (15) should hold by construction as long as the shock remains in a reasonable interval, as is the case in this model.

good *h* produced in country *u*. Define $\tau_{u,a}$ and $\tau_{u,h}$ as the ad-valorem tariffs faced by country *u* when importing low-differentiation products and high-differentiation products, respectively. The script *i* can be omitted since having a continuum of households $i \in [0, 1]$ populating the country implies that the consumption of the representative household *i*, denoted by $c_{1,i}(i)$, is also the aggregate consumption in the country, which is denoted by $c_{1,i}$. I derive the optimal prices from the point of view of a firm in country 1. The market-clearing condition $c^u_{1,i}(h) = y^u_{1,i}(h)$ is satisfied. A firm in country 1 distributing the foreign good *h* in its own market solves the following profit-maximization problem

$$\max_{y_1^2(h)} p_{1,t}^2(h) y_{1,t}^2(h) - \widetilde{p_{1,t}^2}(h) (1+\tau_{1,h}) y_{1,t}^2(h) - \delta p_{1,n,t} \ y_{1,t}^2(h).$$
(16)

Due to the perfect competition that these firms face, it is easy to see that the firms' optimal prices are equal to their marginal costs. Thus, following the same strategy in the low-differentiation sector, the consumer import prices in country 1 can be written as

$$p_{1,t}^{2}(h) = \widetilde{p_{1,t}^{2}}(h)(1+\tau_{1,h}) + \delta p_{1,n,t},$$

$$p_{1,t}^{2}(a) = \widetilde{p_{1,t}^{2}}(a)(1+\tau_{1,a}) + \delta p_{1,n,t}.$$
(17)
(18)

The consumers can buy goods h and a that are produced either in country 1 or in country 2. Then, the distribution sector in country 1 will import the foreign goods that are priced lower than the domestic goods, that is, when

$$p_{1,t}^2(q) < p_{1,t}^1(q), \ q = h, a.$$

Note that when the domestic firms in the tradable sectors sell their products in their own markets, they sell the products directly to the consumers. That is, no distribution firm is involved in the selling process. Therefore, the consumer prices correspond to the producer prices:

$$p_{1,t}^{1}(h) = \widetilde{p_{1,t}^{1}}(h), \tag{19}$$

$$p_{1,t}^1(a) = \widetilde{p_{1,t}^1}(a).$$
 (20)

Analogous expressions apply to country 2.

3.2.3 Tradable sectors

Given the demands $c_{1,l}^{1}(h)$ and $c_{2,l}^{1}(h)$, the representative firm in country 1 in the sector producing high-differentiation goods chooses $\tilde{p}_{1,l}^{1}(h)$ and $\tilde{p}_{2,l}^{1}(h)$ to maximize its profit. The firm is a monopolistic supplier of the good *h* and faces the whole demand for the good *h*; hence, $y_{1,l}^{1}(h) = c_{1,l}^{1}(h)$ and $y_{2,l}^{1}(h) = c_{2,l}^{1}(h)$. The problem therefore is

$$\underset{\widetilde{p_{1,t}^{(1)}(h),\widetilde{p_{1,t}^{(2)}(h)}}{Max}}{Max} \widetilde{p_{1,t}^{(1)}(h)} y_{1,t}^{(1)}(h) + \mathcal{E}_{t} \widetilde{p_{2,t}^{(1)}(h)} y_{2,t}^{(1)}(h) - W_{1,t} l_{1,H,t}$$
(21)

subject to

$$\begin{aligned} c_{1,t}^{1}(h) &= \left(\frac{\alpha I_{1,t}}{\widetilde{p_{1,t}^{1}}(h)}\right)^{\theta} \frac{1}{C_{1,H,t}^{\theta-1}} \quad , c_{2,t}^{1}(h) = \left(\frac{\alpha I_{2,t}}{\widetilde{p_{2,t}^{1}}(h)(1+\tau_{2,h}) + \delta p_{2,n,t}}\right)^{\theta} \frac{1}{C_{2,H,t}^{\theta-1}}, \\ y_{t}^{1}(h) &= \frac{l_{1,H,t}}{b_{1}(h)} = y_{1,t}^{1}(h) + y_{1,t}^{2}(h). \end{aligned}$$

This problem is based on the assumption that the tradable firm knows the foreign demand but does not know the prices chosen by the foreign competitors. The intuition underlying this assumption is that the tradable firms do not know the decisions made by the competitors since they are in a different country. However, the firm knows the demand through an intermediary, the foreign distribution firm.

Using (17) and (19), I can write the optimal producer prices of the high-differentiation tradable good $\widetilde{p_{1}}(h)$ and $\widetilde{p_{2}}(h)$ as

$$\widetilde{p_{1,t}^{0}}(h) = \frac{\theta}{\theta - 1} W_{1,t} b_1(h),$$
(22)

$$\widetilde{p_{2,t}^{1}}(h) = \frac{\theta}{\theta - 1} \left[\frac{1}{\mathcal{E}_t} + \frac{1}{(1 + \tau_{2,h})} \frac{\delta}{\theta} \frac{p_{2,n,t}}{W_{1,t}b_1(h)} \right] W_{1,t}b_1(h).$$
(23)

Similarly, the optimal producer prices of the low-differentiation tradable good $\widetilde{p_{1,t}}(a)$ and $\widetilde{p_{1,t}}(a)$ are given as

$$\widetilde{p_{1,t}^{1}}(a) = \frac{\psi}{\psi - 1} W_{1,t} b_1(a),$$
(24)

$$\widetilde{p_{2,t}^{1}}(a) = \frac{\psi}{\psi - 1} \left[\frac{1}{\mathcal{E}_t} + \frac{1}{(1 + \tau_{2,a})} \frac{\delta}{\psi} \frac{p_{2,n,t}}{W_{1,t}b_1(a)} \right] W_{1,t}b_1(a).$$
(25)

The need for local nontraded goods in the foreign market implies that the elasticities of the demands for the traded goods h and a at the wholesale level will be different across countries. The firms then charge different prices in the home market and in the foreign market as shown above. The empirical evidence and the theoretical papers also support this price determination (see Devereux and Engel, 2003 and Choudhri, Faruquee and Hakura, 2005).

3.2.4 Nontradable sector

A competitive domestic firm in country 1 producing nontradable goods solves the problem

 $\operatorname{Max} p_{1,t}(n) y_{1,t}(n) - W_{1,t} l_{1,N,t} \text{ subject to } y_{1,t}(n) = l_{1,N,t}.$ (26)

Since the firms in this sector face perfect competition, their optimal price is given by

 $p_{1,t}(n) = p_{1,n,t} = W_{1,t}$ (27)

A similar price is chosen by the nontradable firms in country 2.

3.3 Government

It is assumed that the government runs a balanced budget in each period, and that it does that by transferring resources directly to the consumers. The revenues from the import tariffs transferred to the households in country 1 and in country 2 are given by

$$RT_{1,t} = \int_{a_{1,t}}^{1} \tau_{1,a} \widetilde{p}_{1,t}(a) c_{1,t}(a) da + \int_{h_{1,t}}^{1} \tau_{1,h} \widetilde{p}_{1,t}(h) c_{1,t}(h) dh,$$
$$RT_{2,t} = \int_{0}^{a_{2,t}} \tau_{2,a} \widetilde{p}_{2,t}(a) c_{2,t}(a) da + \int_{0}^{h_{2,t}} \tau_{2,h} \widetilde{p}_{2,t}(h) c_{2,t}(h) dh.$$

The revenues are calculated over the set of goods where there is trade between the two countries. The limits of the integrals represent the area where this occurs. This is described in the following section.

The government controls the money supply. Both the revenues from the import tariffs and the seigniorage revenues are rebated to the public in the form of lump-sum transfers. No government spending is assumed. The transfers made by the government in country 1 and in country 2 to their representative households i and j, respectively, are given as follows

$$T_{1,i}(i) = M_{1,i} - M_{1,i-1} + RT_{1,i},$$
(28)

$$T_{2}(j) = M_{2} - M_{2-1} + RT_{2}.$$
 (29)

I assume that

 $E(M_{1,t+1}) = M_t.$

An analogous expression holds for the expected money supply in country 2. This expectational assumption on money supply states that in the present period *t*, the household expects that monetary policy will not change in the next period. In other words, an increase in the money supply is a non-expected increase. This assumption allows us to have ERPT in the response to a change in money supply as the prices and the exchange rate adjust to the new monetary conditions, but not the wages because they have rigidity and have been determined taking into account the money supply in the previous period. In the case where one enjoys perfect foresight over the monetary policy, the wages in the next period would be determined given the future monetary conditions and therefore wages would adjust along with other variables such as prices. Since the pass-through arises due to the wage rigidity, this adjustment in wages would eliminate the ERPT as wages adjust in the direction opposite to that of the prices. Uncertainty about the future monetary conditions seems a realistic assumption.

3.4 Equilibrium

Dropping the scripts *i* and *j*, the variables can be interpreted in per-capita or aggregate terms. Given the initial holdings of bonds $(B_{1,0} \text{ and } B_{2,0})$ and money $(M_{1,0} \text{ and } M_{2,0})$, the equilibrium for this economy is a collection of the allocations for country 1 $(c_{1,t}(a), c_{1,t}(h), c_{1,t}(n), l_{1,t}, l_{1,H,t}, l_{1,A,t}, l_{1,N,t}, y_{t}^{1}(a), y_{t}^{1}(h), \text{ and } y_{1}^{1}(n))$, the allocations for country 2 $(c_{2,t}(a), c_{2,t}(h), c_{2,t}(n), l_{2,t}, l_{2,H,t}, l_{2,A,t}, l_{2,N,t}, y_{t}^{2}(a), y_{t}^{2}(h), \text{ and } y_{t}^{2}(n))$, the prices in country 1 $(p_{1,t}^{1}(h), p_{1,t}^{2}(a), p_{1,t}^{2}(a), \text{ and } p_{1,n,t})$, the prices in country 2 $(p_{1,t}^{1}(h), p_{2,t}^{2}(h), p_{1,t}^{2}(a), p_{2,t}^{2}(a), \text{ and } p_{2,n,t})$ for $a \in [0, 1]$, $h \in [0, 1]$, t = 0, 1, ..., nominal exchange rate \mathcal{E}_{t} allocations for wages $W_{1,t}$ and $W_{2,t}$ tariff revenues $T_{1,t}$ and $T_{2,t}$, money $M_{1,t}$ and $M_{2,t}$, and $bonds B_{1,t}^{1}, B_{1,t}^{2}, B_{1,t}^{1}, B_{2,t}^{1}$ for all t, the equilibrium is such that (i) taking as given the prices, the consumers' problem (1) subject to (2) is satisfed; (ii) taking as given the demands, the domestic firms engaged in the distribution of high-differentiation tradables, low-differentiation tradables, and nontradables solve problems (16), (21), and (26), respectively, and the foreign firms solve analogous problems; (iii) the prices are given by

$$\begin{split} p_{1,t}(h) &= Min[\widetilde{p_{1,t}^{1}}(h), \qquad \widetilde{p_{1,t}^{2}}(h)(1+\tau_{1,h}) + \delta p_{1,n,t}], \\ p_{2,t}(h) &= Min[\widetilde{p_{2,t}^{1}}(h)(1+\tau_{2,h}) + \delta p_{2,n,t}, \qquad \widetilde{p_{2,t}^{2}}(h)], \\ p_{1,t}(a) &= Min[\widetilde{p_{1,t}^{1}}(a), \qquad \widetilde{p_{1,t}^{2}}(a)(1+\tau_{1,a}) + \delta p_{1,n,t}], \\ p_{2,t}(a) &= Min[\widetilde{p_{2,t}^{1}}(a)(1+\tau_{2,a}) + \delta p_{2,n,t}, \qquad \widetilde{p_{2,t}^{2}}(a)]; \end{split}$$

(iv) the following market-clearing conditions hold

$$\begin{split} &\int_{0}^{1} l_{1,H,t} dh + \int_{0}^{1} l_{1,A,t} da + l_{1,N,t} = l_{1,t}, \\ &\int_{0}^{1} l_{2,H,t} dh + \int_{0}^{1} l_{2,A,t} da + l_{2,N,t} = l_{2,t}, \\ &y_{t}^{1}(h) + y_{t}^{2}(h) = c_{1,t}(h) + c_{2,t}(h), \\ &y_{t}^{1}(a) + y_{t}^{2}(a) = c_{1,t}(a) + c_{2,t}(a), \\ &y_{1,t}(n) = c_{1,t}(n), \\ &y_{2,t}(n) = c_{2,t}(n); \end{split}$$

(v) the transfers (28) and (29) are satisfied; and (vi) the bonds are in zero net supply $B_{1,t}^1 + B_{2,t}^1 = 0$, $B_{1,t}^2 + B_{2,t}^2 = 0$

Substituting the bond price from (4) in (8) yields the following equation:

$$E_t\left(\frac{P_{1,t+1}C_{1,t+1}}{P_{2,t+1}C_{2,t+1}\mathcal{E}_{t+1}}\right) = \frac{P_{1,t}C_{1,t}}{P_{2,t}C_{2,t}\mathcal{E}_t}.$$

Iterating, as in Chari, Kehoe and McGrattan (2002), the previous condition can be rewritten as

$$E_t\left(\frac{P_{1,t}C_{1,t}}{P_{2,t}C_{2,t}\mathcal{E}_t}\right) = \frac{P_{1,0}C_{1,0}}{P_{2,0}C_{2,0}\mathcal{E}_0}.$$

The right-hand side of the previous equation depends on the initial wealth distribution. Let $v = \frac{P_{1,0}C_{1,0}}{P_{2,0}C_{2,0}\mathcal{E}_0} > 0$. Since symmetrical initial conditions are assumed, v = 1. This implies that

$$\mathcal{E}_t = \frac{P_{1,t}C_{1,t}}{P_{2,t}C_{2,t}}$$

There are two possible non-stochastic steady states, one when there is no trade and the other when there is trade across countries. First, there is no trade in equilibrium when $W_1(\gamma+\lambda) < W_2 \mathcal{E}(1+\tau_{1,h})\gamma+\delta W_1$ and $W_2(\gamma+\lambda) < \frac{W_1}{\mathcal{E}}(1+\tau_{2,h})\gamma+W_2$. Second, if $W_1(\gamma+\lambda) > W_2 \mathcal{E}(1+\tau_{1,h})\gamma+W_1$ and $W_2(\gamma+\lambda) > \frac{W_1}{\mathcal{E}}(1+\tau_{2,h})\gamma+\delta W_2$, then the pattern of trade, production, and specialization in equilibrium looks as follows.

Graphic 5 Pattern of trade, production, and specialization



Country 1 produces in the interval $[1, h_1]$ and exports the goods in the interval $[0, h_2]$. Country 2 produces in the interval $[h_2, 1]$ and exports the goods in the interval $[h_1, 1]$. Therefore, country 1 imports the goods in the interval $[h_1, 1]$ and country 2 imports the goods in the interval $[0, h_2]$. The goods in the interval $[h_2, h_1]$ are not traded.

$$h_{1} = \frac{W_{2}\mathcal{E}(1+\tau_{1,h})(\gamma+\lambda) + \delta W_{1} - \gamma W_{1}}{W_{1}\lambda + W_{2}\mathcal{E}(1+\tau_{1,h})\lambda}.$$

$$h_{2} = \frac{W_{2}\mathcal{E}(\gamma+\lambda) - \delta \mathcal{E}W_{2} - \gamma W_{1}(1+\tau_{2,h})}{W_{1}\lambda(1+\tau_{2,h}) + W_{2}\mathcal{E}\lambda}.$$
(30)

The prices and production patterns by the variety are included in appendix 2.

In the steady state, the monetary stances are given by $\frac{W_1}{W_2} = 1$, and since we have symmetry, by $\frac{W_1}{W_2} = 1$. Further, the steady state bonds holdings are assumed to be zero, B_1 , $B_2 = 0$. The equations summarizing the solution of the model in the steady state are included in appendix 3.

3.5 ERPT by good

3.5.1 ERPT at the producer level

Logistics costs, tariffs, and the market structure of each sector have important implications for the ERPT level at the producer level. By definition, the ERPT for the high-differentiation goods is given by

$$\frac{\frac{\partial p_{1,t}^1(h)}{\tilde{p_{2,t}^1(h)}}}{\frac{\partial \mathcal{E}_t}{\mathcal{E}_t}} = \frac{1}{1 + \frac{\mathcal{E}_t}{(1 + \tau_{2,h})} \frac{\delta}{\theta} \frac{W_{2,t}}{W_{1,t}} \frac{1}{b_1(h)}}.$$
(32)

The ERPT for the low-differentiation goods is given by

$$\frac{\frac{\partial p_{2,t}^1(a)}{p_{2,t}^2(a)}}{\frac{\partial \mathcal{E}_t}{\mathcal{E}_t}} = \frac{1}{1 + \frac{\mathcal{E}_t}{(1 + \tau_{2,a})} \frac{\delta}{\psi} \frac{W_{2,t}}{W_{1,t}} \frac{1}{b_1(a)}}.$$
(33)

Note that the pass-through arises due to the wage rigidities; if there were no wage rigidities, the pass-through would be zero since the wages would make adjustments in the direction to opposite to that of the prices. Eliminating the nontradable sector, δ = 0, would cause complete the pass-through since the firm would be pricing the good for export in the domestic currency, but incorporating the taxes. This is because all the costs would be domestic.

To gain a more intuitive understanding about the previous expressions, consider the producer price elasticity of the export demand of good h

$$\frac{\frac{\partial c_{2,t}^{1}(h)}{\partial \overline{p_{2,t}^{1}(h)}}}{\frac{\partial \overline{p_{2,t}^{1}(h)}}{p_{2,t}^{1}(h)}} = \theta \frac{\widetilde{p_{2,t}^{1}(h)}(1+\tau_{2,h})}{\widetilde{p_{2,t}^{1}(h)}(1+\tau_{2,h})+\delta p_{2,n,t}} = \frac{\theta(1+\tau_{2,h})W_{1,t}b_{1}(h) + \mathcal{E}_{t}\delta W_{2,t}}{(1+\tau_{2,h})W_{1,t}b_{1}(h) + \mathcal{E}_{t}\delta W_{2,t}}$$

The producer price elasticity measures the effect of a decision of a foreign firm in terms of its export prices on the final demand. It is worth stressing that the parameters that are positively related to high price elasticity such as θ , $\tau_{2,h}$ and $b_1(h)$ are also positively related to high ERPT. The intuition behind this is that the firm moves the price in response to an exchange rate movement to maintain its profits to the extent that the demand responds to its price decisions. In other words, the higher the sensitivity of the consumption to the changes in the producer

prices, the higher the incentive of the firm to move the prices in response to the exchange rate fluctuations in order to balance its profits.

Equations (32) and (33) indicate that the low-differentiation products face higher levels of pass-through than the high-differentiation products. This occurs because of the following two reasons:

a) Homogeneity of the good: Since $\psi > \theta > 1$, the low-differentiation goods have higher levels of pass-through than the high-differentiation goods. To gain an intuitive understanding of this result, one can compare a firm with perfect competition, that is, $\psi \rightarrow \infty$ (homogeneous good), with a firm with market power. In the case of perfect competition, the ERPT would be complete since the firms have no other choice than to move the price due to the movements in the exchange rate; otherwise, they would face losses and be out of the market. In contrast, a firm with market power sets the price of its product as high as possible given the consumers? preference for the firm's product. This implies that the producer firm has already set the highest possible price to maximize profits (higher than the marginal cost and taking into account the consumers' preference and income). Therefore it is not profitable to move the price by the same degree as the exchange rate, particularly when the producer firm does not have total control over the final price of the good since there is a local cost associated with its distribution to the final consumer. The fact that the final price is given by both production and distribution costs reduces the capacity of the foreign producer firm to balance its profits after an exchange rate movement by making changes to the prices as it does not fully control the final price, leading to incomplete pass-through. There is empirical evidence regarding industrial organization that supports these results. For example, Campa and Goldberg (2005) say that "Overall, we again find that partial pass-through is a common phenomenon particularly among heterogeneous products. More homogenous products have more extreme pass-through values." Microeconomic studies also show that firms with higher market power move the prices less due to the movements in the exchange rate. Examples are De Stefano (2004) on the Italian car market and Kardasz and Stollery (2005) on the Canadian manufacturing industry.

In addition, the recent study of Bugamelli and Tedeschi (2008) finds that the ERPT is lower in oligopolistic industries than in competitive ones and (as the model in this paper suggests) that the reason for this is the pricing-to-market behavior of firms with pricing power.

b) Tariffs: The varieties in the sector with higher tariffs present higher levels of pass-through at the producer level and at the consumer level (as will be shown later). This result is supported by recent empirical evidence. Mallick and Margues (2008) show that import prices react more to exchange rate changes in sectors with higher effective protection rates. The reasons behind this result are not explored in this study, but the authors suggest that import restrictions or high tariffs might reduce the capacity of foreign firms to maintain market share by adjusting their profit margins in response to a change in the exchange rate. The model developed in this paper provides an alternative explanation for the relation between tariffs and pass-through at the sector level. The sector with higher tariffs has a higher producer price elasticity of the export demand, implying that a foreign firm in a sector with high tariffs has more space to compensate the change in its profits resulting from an exchange rate movement by changing its price as the demand is more producer price sensitive. Empirical evidence shows that agricultural goods and raw materials, the goods with higher levels of pass-through, face higher tariffs than manufacturing goods. For example, according to the CPII, in the European Union (EU)-15, the equivalent average tariff in 2001 in the agricultural sector and in the manufacturing sector is 15.0% and 1.8%, respectively. The calculations from the IMF-World Bank (2001) give an average tariff of 19.5% and 4.1%, respectively. In general, the difference in the pass-through levels between these two sectors comes from the differences in the monopoly power and tariffs.

3.5.2 ERPT at the consumer level

The decision to move the price is only the decision of the firms producing tradables goods, as the competitive distribution firm only buys the foreign tradable product at the given price in its own currency, pays its import taxes, and distributes it to the consumers using the nontradable good. This means that the level of ERPT at the consumer level is a share of the ERPT at the producer level since the price at the consumer level contains the price at the producer level.

The difference between the pass-through at the producer level and at the consumer level is represented by the market structure of each sector. By definition, the ERPT for the high-differentiation goods sector at the consumer level is given by

$$\frac{\frac{\partial p_{2,t}^1(h)}{p_{2,t}^1(h)}}{\frac{\partial \mathcal{E}_t}{\mathcal{E}_t}} = \frac{1}{1 + \frac{\mathcal{E}_t \delta}{(1 + \tau_{2,h})} \frac{W_{2,t}}{W_{1,t}} \frac{1}{b_1(h)}}.$$
(34)

By the same token, the ERPT for the low-differentiation goods sector is

$$\frac{\frac{\partial p_{1_t}^1(a)}{p_{1_t}^2(a)}}{\frac{\partial \mathcal{E}_t}{\mathcal{E}_t}} = \frac{1}{1 + \frac{\mathcal{E}_t \delta}{(1 + \tau_{2,a})} \frac{W_{2,t}}{W_{1,t}} \frac{1}{b_1(a)}}.$$
(35)

Note that at the consumer level, the presence of tariffs yields that the low-differentiation products face higher levels of pass-through than high-differentiation products. Further, the goods produced with low technology (high $b_4(q)$, q = h, a) face higher pass-through levels.

The aggregate ERPT is obtained by calculating the pass-through from the aggregate price indexes. The aggregate pass-through will depend on the number of varieties imported by sector and the pass-through level of each variety. It is expected that an FTA will result in an increase in the number of varieties imported and a decrease in the pass-through level of each variety. Then, the relative increase in the imports of one sector with respect to the other together with the new pass-through level by sector modifies the aggregate pass-through level.

The analytical expression obtained by calculating the aggregate pass-through does not clearly illustrate the total effect. Thus, some numerical exercises are needed.

4 Calibration and Parametrization

The model has been calibrated to the USA economy for the year 1993, when the USA began signing FTAs with different countries. In order to classify the industries according to the product differentiation, I follow Oliveira, Scarpetta and Pilat (1996). These authors calculate the mark-ups for the period 1970-1992, and classify the industries into high-differentiation industries and low-differentiation industries. Most of the industries classified by Oliveira, et al. (1996) as low-differentiation industries are those producing commodities such as petroleum, leather, wood, steel, etc. The industries classified as high-differentiation industries include machinery and equipment, motor vehicles, medicines, etc. The USA mark-ups for each group are

$$\frac{\psi}{\psi-1} = 1.12$$
 and $\frac{\theta}{\theta-1} = 1.33$.

The low-differentiation goods' share in consumption $1 - \alpha$ is chosen such that 40% of the consumption expenditure is devoted to low-differentiation goods. This is consistent with the share of the low-differentiation goods (commodities) in the USA CPI for the base year 1993. According to the World Trade Organization (WTO, 2004), in 2002, the average USA tariffs in the agricultural and nonagricultural sectors were 6.9% and 3.2%, respectively. The exercise involves reducing the average tariffs in both sectors from 20% (the average USA tariffs in 1980) to their current levels in 2002. These values do not seem to be unrealistic since institutions such as the World Bank and the Food and Agriculture Organization (FAO) argue that while average tariffs have generally been reduced following the Uruguay Round of trade negotiations, high duties continue to be placed on certain agricultural products in order to protect domestic producers.

The benchmark for the number of nontraded goods required to distribute a tradable good, δ , is chosen as follows. Burstein, Neves and Rebelo (2003) estimate that the distribution costs represent more than 50% of the retail price in all agricultural products. These margins range from 54% (eggs) to 82% (fresh fruits). The total distribution margin in the USA is estimated as 42%. Then, I set δ such that the share of the nontraded goods on the final price in both sectors ranges from 42% to 82%, depending on the variety, given δ = 1.403. The parameter on the disutility of labor, *k*, is set equal to 1 to restrict the labor-output ratio to 2/3. The mark-up over the marginal cost of labor is chosen as $\frac{\varphi}{\varphi-1}$ = 1.15. This mark-up is consistent with the estimates of the mark-up of union wages over nonunion wages estimated by Lewis (1986).

The productivity parameters γ and λ are obtained after some manipulations over equations (30) and (31). By doing this, I obtain two equations that depend only on wages, tariffs, and marginal costs, which can be calculated or approximated from the data. An explanation of this process is included in appendix 4. The parameters obtained are $\gamma = 0.076$ and $\lambda = 2.110$.⁷

⁷ The restriction $\lambda > \gamma \tau_{1,a} + \delta$ must hold to ensure trade across the countries in both sectors.

5 Quantitative Exercise

In this section, I perform two simple exercises. The first one is the determination of how the ERPT by variety is affected by the tariffs elimination. In the second exercise, the effects of an FTA on the aggregate pass-through level are analyzed. Specifically, I compute the impact effects of nominal monetary shock (which is defined as a temporary 1% deviation from the steady-state value) on the pass-through by variety and by sector, on the aggregate pass-through, and on the intervals of the varieties traded in two different scenarios: one where there are high tariffs and another where the tariffs are low.

A 1% increase in M_1 from the steady-state causes a movement on the exchange rate from the steady state, and this, in turn, affects the import prices: ERPT.⁸ Since wages take one period to adjust, a new steady state is reached after the shock. The one-period-in-advance wage setting is a shortcut that allows the incorporation of wage rigidity and an analytical solution for the model. This wage setting also has a disadvantage. It does not permit the analysis of some interesting dynamics such as the volatility in macroeconomic variables and the ERPT over time. In spite of this disadvantage, this wage setting seems adequate considering that the model focuses on price determination and trade.

To analyze how the different variables change due to a 1% deviation in the money supply from the steady state, I use the method of undetermined coefficients described in Uhlig (1997). Using this method, the deviations of the endogenous variables from the steady state can be rewritten as linear functions of the deviation of the money supply from the steady state.

Graphic 6 shows the results of the first exercise, that is, the ERPT by variety under the two scenarios: high and low tariffs. A decrease in the ERPT in each variety is observed as a consequence of the trade liberalization. High-differentiation goods exhibit a higher decrease in the ERPT by good than the low-differentiation goods because of the higher tariffs elimination in this sector. Further, note that in general, the varieties of the low-differentiation sector reach higher levels of pass-through than the varieties of the high-differentiation sector. This is due to the higher tariffs and the lower market power of the low-differentiation sector. As suggested by the equations (32) and (33), the varieties with the lowest technology in country 1 are those with the highest pass-through.



Graphic 6 Computational results (ERPT by good under different trade liberalization scenarios)

⁸ The shocks to the velocity (x) and the disutility of labor (k) are not considered in the model because the exercise focuses on the ERPT in two isolated scenarios, high and low tariffs. For this purpose, it is preferred that a simple mechanism be used to yield the ERPT: an increase in the money supply. This simplification does not alter the general result since the shocks, in the case they were to be incorporated, would be present in both scenarios. However, these shocks could have some influence on the magnitudes of the ERPT since Corsetti and Dedola (2005) finds that these shocks seem to be relevant in modeling the ERTP.

Chart 6 contains the intervals of the varieties imported and the ERPT by sector, and the aggregate pass-through. The first column presents the results in the absence of an FTA while the second column presents the results under an FTA. The third column shows the percentage change in each variable between the two scenarios. The quantitative results support our expected results. First, in the first column, we can see that in the absence of an FTA, for the high-differentiation sector, the model displays an ERPT on the producer import prices of 0.3718, as compared to 0.4519 in the low-differentiation sector. That is, at the producer level, the high-differentiation goods are those with the lowest pass-through level. The opposite occurs at the consumer level because the market power is not the driving force anymore. At the consumer level, the difference in the pass-through between the goods h and a is small, as the difference is given only by the tariffs. As a consequence of the trade liberalization, the varieties of the high-differentiation goods that are imported are more than the varieties of the low-differentiation goods that are imported. The new varieties imported are those with a higher ERPT since they have a lower productivity level, causing a higher aggregate ERPT at the consumer level in the high-differentiation sector. On the aggregate, the model shows an ERPT at the producer level of 0.4038 under high tariffs and 0.3820 under low tariffs. These results are in line with the ERPT in the USA for the period 1994-2003, which is 0.42.9 The passthrough levels by sector are also close to the results found by Campa and Goldberg (2005) for the OECD countries: 0.50% in the manufacturing sector and 0.64% in the non-manufacturing and raw materials sectors.

Chart 6

Compu	tational	results	(selected	variables	under	different	trade	libera	lization	scenarios
-------	----------	---------	-----------	-----------	-------	-----------	-------	--------	----------	-----------

	No FTA (1)	FTA (2)	Percent change (2) - (1)
High-differentiation goods sector			
Varieties imported:	[0.8509,1.000]	[0.8356,1.000]	10.27%
ERPT:			
Producer import price	0.3718	0.3511	-5.55%
Consumer import price	0.1520	0.1417	-6.84%
Low-differentiation goods sector			
Varieties imported:	[0.8509,1.000]	[0.8392,1.000]	7.86%
ERPT:			
Producer import price	0.4519	0.4283	-5.21%
Consumer import price	0.1342	0.1270	-5.35%
Aggregate			
ERPT:			
Producer import price	0.4038	0.3820	-5.40%
Consumer import price	0.1449	0.1358	-6.28%

The model also predicts that the sign of the FTA would cause a drop in the pass-through at the producer level in the high-differentiation sector of 5.55% and in the low-differentiation sector of 5.21%. At the consumer level, the decrease in the high-differentiation sector is 6.84% and that in the low-differentiation sector 5.35%. In addition, the aggregate pass-through decreases at the producer level by 5.4% and at the consumer level by 6.28%. These measures are located between the values of the two sectors because after the FTA, the varieties of goods imported increase in both the sectors.

⁹ The estimation is based on the estimation method proposed by Campa and Goldberg (2005):

$$\Delta imp_t = \alpha + \gamma \Delta imp_{t-1} + \sum_{i=0}^{-} \beta_i \Delta ner_{t-i} + \sum_{i=0}^{-} \delta_i \Delta mc_{t-i} + u_t,$$

where Δ represents the first differences, *imp* is the import price index (in logarithms), *ner* denotes the effective nominal exchange rate in the foreign currency per US dollars (in logarithms), *mc* is the marginal cost (in logarithms), and *u* is the error term. The marginal cost, *mc*, is the logarithm transformation of MC=(NER/RER)*ULC, where NER is the effective nominal exchange rate, RER is the effective real exchange rate, and ULC denotes the unit labor costs of the non-agricultural sector (the only sector for which the data are available). The measurement of the marginal cost is based on Otani, Shiratsuka and Shirota (2003). The ERPT is given by β_0 . The sources of the data are included in appendix 1. The varieties of high-differentiation goods increase by 10.27% and those of low-differentiation goods increase by only 7.86%. Then, the aggregate ERPT depends on both the new import bundle and the decrease in the pass-through by good. This result sheds light on the reason why the pass-through level has been decreasing over time. McCarthy (2007) and Otani, Shiratsuka and Shirota (2003) document this fact. In particular, Otani et al. (2003) suggest that the decrease in the pass-through level in Japan is due to its trade liberalization, which is in line with the results obtained here.

In this model, the parameter λ plays an important role since it affects both the ERPT by product and the number of varieties imported. A sensitivity analysis on this parameter is done. Chart 7 contains the varieties imported, and the aggregate ERPT at both the producer level and the consumer level under both tariffs levels for different values of λ :1.60 and 4.00 (that is, one below and one above the benchmark of 2.11). Note that under both values of λ , the condition $\lambda > \gamma \tau_{1a} + \delta$ holds, which ensures trade across countries in both sectors.

Chart 7 ERPT under different scenarios of the parameter of technology λ

									Percen	t change
	No FTA (1)			FTA (2)				(2) - (1)		
	Varieties of	Varieties of	ERPT	ERPT	Varieties of	Varieties of	ERPT	ERPT	ERPT	ERPT
	imported low-	imported high-	producer	consumer	imported low-	imported high-	producer	consumer	producer	consumer
	differentiation	differentiation	import	import	differentiation	differentiation	import	import	import	import
	goods	goods	price	price	goods	goods	price	price	price	price
Lambda=1.60	0.0516	0.0516	0.3542	0.0972	0.0598	0.0579	0.3332	0.0902	-5.91%	-7.12%
Lambda=2.11	0.1491	0.1491	0.4038	0.1449	0.1643	0.1607	0.3820	0.1358	-5.40%	-6.28%
Lambda=4.00	0.2934	0.2934	0.4314	0.1873	0.3192	0.3132	0.4102	0.1773	-4.91%	-5.32%

When there is no an FTA the varieties imported go from 5 percent to 29 percent of total depending on the level of lambda. The ERPT at producer level goes from 35 to 43 percent while at consumer level it goes from 9 to 18 percent. Similar variations arise under an FTA. The increase in lambda is a decrease in the productivity level which causes that each of the goods the country is importing has a higher pass-through. The percent change of the aggregate ERPT due to the trade liberalization does not change that much across scenarios. The fall of ERPT goes from 4.91 to 5.91 percent at producer level and from 5.32 to 7.12 percent at consumer level.

The model correctly displays the idea of the paper: a decrease in the ERPT by good and on aggregate due to the tariffs elimination and the presence of a new import bundle. However, the decrease in the ERPT does not seem to be very high. There are two reasons for this. First, the symmetry of the model does not allow high movements on the varieties of the imported goods. Second, the tariffs elimination is the only instrument resulting in a decrease in ERPT; however, at the same time, the trade liberalization is also characterized by the elimination of high nontariff barriers that are not considered in this model. In general, the non-tariff barriers represent the additional regulations that add to the final price of a product as an additional cost. Thus, these non-tariff barriers can be modeled as a tax over the producer price, much like in the model. I conduct an exercise to take into account these non-tariff barriers. According to Looi, Nicita and Olarreaga (2004), in the USA, the non-tariff barriers, measured as an ad-valorem equivalent, are two times the average tariff. Consequently, the new simulation reducing the average tariffs in both sectors from 60% (which is the sum of the average tariffs in 1980, given as 20%, and the non-tariff ad-valorem equivalent, given as 40%) to their current levels in 2002. The 60% figure seems to be moderate, considering that before the trade liberalization in Mexico and the USA, the non-tariff barriers were guite heavy.

Chart 8

Computational results when non-tariff barriers are includes (selected variables)

	No FTA, tariff and non-tariff barriers (1)	FTA (2)	Percent change (2)-(1)
High-differentiation goods sector			
Varieties imported:	[0.8795,1.000]	[0.8356,1.000]	36.43%
ERPT:			
Producer import price	0.413	0.3511	-14.99%
Consumer import price	0.173	0.141	-18.02%
Low-differentiation goods sector			
Varieties imported:	[0.8795,1.000]	[0.8392,1.000]	33.44%
ERPT:			
Producer import price	0.513	0.4283	-16.52%
Consumer import price	0.154	0.1270	-17.35%
Aggregate			
ERPT:			
Producer import price	0.453	0.3820	-15.67%
Consumer import price	0.165	0.1358	-17.77%

Chart 8 is constructed in the same manner as chart 6; however, the results under the label No FTA correspond to the new simulation. The results show that when the non-tariff barriers are included, the ERPT decreases by 16% at the producer level and by 18% at the consumer level. These are substantial decreases in the ERPT, in particular, considering that there are many other economies that have adopted stronger liberalization policies.

6 Conclusion

This paper investigates the effect of the trade liberalization on the ERPT. To do so, it employs an empirical estimation of the effects of the NAFTA on the Mexican ERPT, and uses a Ricardian general equilibrium model that incorporates tradable goods, which may have high or low substitutability, a nontradable sector, and a distribution sector in each of the two symmetric countries.

The main contribution of the model proposed in this paper lies in a disaggregated analysis of the effect of the trade liberalization on the ERPT. An analysis of this type requires limiting some other features of the model, in particular, in terms of the dynamics such as the ERPT over time and the volatility of the macroeconomic variables.

In particular, the model identifies two channels that explain how the bilateral trade liberalization alters the ERPT. The first channel is the direct relationship between the tariffs and the pass-through by good. The second channel is the effect that tariffs have on the composition of imports, altering indirectly the aggregate pass-through.

With respect to the first channel, the model shows that the reduction in trade tariffs decreases the pass-through level by good since the tariffs are positively related to the ERPT. The intuition is that the tariffs increase the proportion of the final price influenced by the foreign firm (gross export price). As the foreign firms have more influence on the final price, in terms of amount, they have more incentives to pass-through a higher proportion of the exchange rate movements to the import prices since the firm balances its profits with price movements.

The second channel is simpler. As the high-differentiation goods face a more stringent tariffs elimination, its share in total imports increases. Since the high-differentiation goods have a lower ERPT than the low differentiated goods, the aggregate ERPT falls. More homogeneity in the goods implies more competition and more pass-through. The higher the competition, the closer the price is to its marginal cost. This implies that after an exchange rate movement, a competitive firm does not have any option but to move the price of its product; otherwise, it would face losses.

In addition, an analysis of the effect of the NAFTA on the ERPT to Mexican import prices is performed, concluding that the trade liberalization has decreased the pass-through level in Mexico.

This paper opens more questions about the effect of the trade liberalization on the ERPT, for example, how and to what extent subsidies affect the ERPT. The case of export subsidies is particularly interesting because they are highly concentrated in some raw materials and agricultural products, which are already seeing intensive discussions by the monetary authorities owing to their high price volatility.

References

[1] Anderson, Theodore and Cheng Hsiao, "Estimation of Dynamic Models with Error Compo-

nents," Journal of the American Statistical Association 76 (1981):598-606.

[2] Auer, Raphael and Thomas Chaney, "Exchange Rate Pass-through in a Competitive model of Pricing-to-market," *Journal of Money, Credit and Banking* 41 (2009):151-175.

[3] Baqueiro, Armando, Alejandro Díaz de León, and Alberto Torres, "Fear of Floating or Fear of Inflation? The Importance of the Pass-through from Exchange Rate to Prices," *Ensayos sobre Política Económica Journal* 44 (2004):64-94.

[4] Barhoumi, Karim, Differences in Long-run Exchange Rate Pass-through into Import Prices in Developing Countries: An Empirical Investigation," *Economic Modelling* 23(2006):926-951.

[5] Bugamelli, Matteo and Roberto Tedeschi, "Pricing-to-market and Market Structure," Oxford

Bulletin of Economics and Statistics 70 (2008):155-180.

[6] Burstein, Ariel, Joao Neves, and Sergio Rebelo, "Distribution Costs and the Real Exchange

Rate Dynamics during Exchange-rate-based Stabilizations," *Journal of Monetary Economics* 50 (2003):1189-1214.

[7] Campa, Jose and Linda Goldberg, "Exchange Rate Pass-through into Import Prices," *Review of Economics and Statistics* 87 (2005):679-690.

[8] Chari, Varadarajan, Patrick Kehoe, and Ellen McGrattan, "Can Sticky Price Models Generate Volatile and Persistent Real Exchange Rates?," *Review of Economic Studies* 69 (2002):533-63.

[9] Choudhri, Ehsan and Dalia Hakura, "Exchange Rate Pass-Through to Domestic Prices: Does the Inflationary Environment Matter?," *Journal of International Money and Finance* 25 (2006):614-639.

[10] Choudhri, Elshan, Hamid Faruquee, and Dalia Hakura, "Explaining the Exchange Rate Pass-through in Different Prices," *Journal of International Economics* 65 (2005):349-374.

[11] Coricelli, Fabrizio, Bostjan Jazbec, and Igor Masten, "Exchange Rate Pass-through in EMU Ac-ceding Countries: Empirical Analysis and Policy Implications," *Journal of Banking and Finance* 30 (2006): 1375-1391.

[12] Corsetti, Giancarlo and Luca Dedola, 2005. "A Macroeconomic Model of International Price Discrimination," *Journal of International Economics* 67 (2005):129-155.

[13] De Stefano, Martino, "Exchange Rate Pass-Through in the Italian Car Market: 1990-1996,"Working Paper, Boston University, 2004.

[14] Devereux, Michael and Charles Engel, "Endogenous Currency of Price Setting in a Dynamic Open Economy Model," NBER working paper 8559, 2001.

[15] Devereux, Michael and Charles Engel, "Monetary Policy in the Open Economy Revisited: Exchange Rate Flexibility and Price Setting Behavior," *Review of Economic Studies* 70 (2003):765-783.

[16] Dornbusch, Rudiger, Stanley Fisher, and Paul Samuelson, "Comparative Advantage, Trade and Payments in a Ricardian Model with a Continuum of Goods," *American Economic Review* 67 (1977):823-839.

[17] Erceg, Christopher and Andrew Levin, Structure and the Dynamic Behavior of the Real Exchange Rate," mimeo, Board of Governors of the Federal Reserve System, 1995.

[18] Goldberg, Pinelopi and Michael Knetter, "Goods Prices and Exchange Rates: What Have We Learned," *Journal of Economic Literature* 35 (1997):1243-1272.

[19] Goldfajn, Ilan and Sergio Werlang, "The Pass-Through from Depreciation to Inflation: A Panel Study," Banco Central do Brasil Working Paper Series No. 5, 2000.

[20] Hansen, Lars, "Large Sample Properties of Generalized Method of Moments Estimators," *Econometrica* 50 (1982):1029-1054.

[21] Looi, Kee, Alessandro Nicita, and Marcelo Olarreaga, "Ad-valorem Equivalents of Nontariff Barriers," The World Bank, 2004.

[22] IMF-World Bank, "Market Access for Developing Countries' Exports," www.worldbank.org/economics/marketaccess.pdf, 2001.

[23] Ito, Takatoshi and Kiyotaka Sato, "Exchange Rate Changes and Inflation in Post-crisis Asian Economies: Vector Autoregression Analysis of the Exchange Rate Pass-through," *Journal of Money, Credit and Banking* 40 (2008):1407-1438.

[24] Kardasz, Stanley and Stollery Kenneth, "Exchange Rate Pass-through in the Canadian Manufacturing: Its Direct and Indirect Components," Applied Economics 37 (2005):1763-1776.

[25] Lewis, Gregg, "Union Relative Wage Effects," in Ashenfelter, Orley and Richard Layard (eds.), *Handbook of Labor Economics*, Vol. 2, Amsterdam: North-Holland, 1986, 1139-1181.

[26] Mallick, Sushanta and Helena Marques, "Pass-Through of Exchange Rate and Tariffs into Import Prices of India: Currency Depreciation versus Import Liberalization," *Review of International Economics* 16 (2008):765-782.

[27] McCarthy, Jonathan "Pass-through of Exchange Rates and Import Prices to Domestic Inflation in Some Industrialized Economies," *Eastern Economic Journal* 33 (2007):511-537.

[28] Mihailov, Alexander, "Exchange Rate Pass-through to Prices in Macrodata: A Comparative Sensitivity Analysis," *International Journal of Finance and Economics* 14 (2009):346-377.

[29] Oliveira, Joaquim, Stefano Scarpetta, Dirk Pilat, "Mark-ups in Manufacturing Industries: Estimates for 14 OECD Countries," OECD Economics Department Working Papers no. 162, 1996.

[30] Otani, A., Shiratsuka, S., Shirota, T., "The Decline in the Exchange Rate Pass-through: Evidence from Japanese Import Prices," *Monetary and Economic Studies* 3 (2003):53-81.

[31] Perron, Pierre, "The Great Crash, the Oil Price Shock, and the Unit Root Hypothesis," Econometrica 57 (1989):1361-1401.

[32] Taylor, John, "Low Inflation, Pass-through and the Price Power of Firms," European Economic Review 44 (2000):1389-1408.

[33] Uhlig, Harald, "A Toolkit for Analyzing Nonlinear Dynamic Stochastic Models Easily," Institute for Empirical Macroeconomics, Federal Reserve Bank of Minneapolis, Updated Version of the Discussion Paper 101 and Updated Version of the Center Discussion Paper 9597, 1997.

[34] WTO, "USA: Country Profile," http://stat.wto.org/CountryPro le/WSDBCountryPFHome. aspx?Language=E, 2004.

[35] Zivot, Eric and Donald Andrews, "Further Evidence on the Great Crash, the Oil Price Shock, and the Unit Root Hypothesis," *Journal of Business and Economic Statistics* 10 (1992):251-270.

Appendix 1. Data sources

Chart 1

Quarterly data: USA		
Series	Description and sources	Sample period
Import price index	IMF International Financial Statistics	Quarterly: 1Q 1994 - 4Q 2003
Nominal effective exchange rate	ominal effective exchange rate IMF International Financial Statistics	
Real effective exchange rate	IMF International Financial Statistics	Quarterly: 1Q 1994 - 4Q 2003
Unit labor costs	Bureau of Labor Statistics (BLS)	Quarterly: 1Q 1994 - 4Q 2003
Quarterly data: Mexico		
Series	Description and sources	Sample period
Import price index	OECD (index calculated from prices	Quarterly: 1Q 1987 - 2Q 2003
Nominal exchange rate	Central Bank of Mexico (Peso/US dollars)	Quarterly: 1Q 1987 - 2Q 2003
Mexico salary index	Central Bank of Mexico	Quarterly: 1Q 1987 - 2Q 2003
USA payroll index	Bureau of Labor Statistics (BLS)	Quarterly: 1Q 1987 - 2Q 2003
Monthly data: Mexico		
Series	Description and sources	Sample period
Import price index	Central Bank of Mexico (index calculated from	Monthly: January 1987 - May 2010
Nominal exchange rate	Central Bank of Mexico	Monthly: January 1987 - May 2010
Mexico manufacturing unit labor costs	Central Bank of Mexico	Monthly: January 1987 - May 2010
USA payroll index	Bureau of Labor Statistics (BLS)	Monthly: January 1987 - May 2010
Data: Explaining the ERPT in Mexico		
Series	Description and sources	Sample period
Manufacturing imports and total imports	INEGI (millons of dollars)	Quarterly: 1Q 1987 2Q - 2003
Exports and imports	INEGI (millons of dollars)	Quarterly: 1Q 1987 2Q - 2003
Mexico inflation	Central Bank of Mexico	Quarterly: 1Q 1987 2Q - 2003
USA inflation	Bureau of Labor Statistics (BLS)	Quarterly: 1Q 1987 2Q - 2003
Mexico GDP	INEGI (millons of dollars)	Quarterly: 1Q 1987 2Q - 2003

Source: own calculations

Appendix 2. Prices and Output

The consumer prices are

$$p_{1}(h) = \begin{cases} p_{1}^{1}(h) = \frac{\theta}{\theta - 1} W_{1}(\gamma + \lambda h) & , h \in [0, h_{1}] \\ p_{1}^{2}(h) = \frac{\theta}{\theta - 1} (W_{2} \mathcal{E}(\gamma + \lambda(1 - h))(1 + \tau_{1,h}) + \delta W_{1}) & , h \in (h_{1}, 1] \end{cases}$$
(1)

$$p_{2}(h) = \begin{cases} p_{2}^{1}(h) = \frac{\theta}{\theta-1} \left(\frac{W_{1}}{\mathcal{E}} (\gamma + \lambda h) (1 + \tau_{2,h}) + \delta W_{2} \right), h \in [0, h_{2}) \\ p_{2}^{2}(h) = \frac{\theta}{\theta-1} W_{2} (\gamma + \lambda (1 - h)) , h \in [h_{2}, 1] \end{cases}$$
(2)

The producer prices are

$$\widetilde{p}_{1}(h) = \begin{cases} \widetilde{p}_{1}^{1}(h) = \frac{\theta}{\theta - 1} W_{1}(\gamma + \lambda h), & h \in [0, h_{1}] \\ \widetilde{p}_{1}^{2}(h) = \frac{\theta}{\theta - 1} (W_{2} \mathcal{E}(\gamma + \lambda(1 - h)) + \frac{\delta}{\theta} \frac{W_{1}}{(1 + \tau_{1})}), h \in (h_{1}, 1] \end{cases}$$
(3)

$$\widetilde{p_2}(h) = \begin{cases} \widetilde{p_2^1}(h) = \frac{\theta}{\theta - 1} \left(\frac{W_1}{\mathcal{E}} (\gamma + \lambda h) + \frac{\delta}{\theta} \frac{W_2}{(1 + \tau_{2,h})} \right), h \in [0, h_2) \\ \widetilde{p_2^2}(h) = \frac{\theta}{\theta - 1} W_2(\gamma + \lambda(1 - h)), \qquad h \in [h_2, 1] \end{cases}$$

$$\tag{4}$$

The production is

$$\begin{split} y^{1}(h) &= \left(\frac{\alpha I_{1}}{p_{1}(h)}\right)^{\theta} \frac{1}{C_{1,H}^{\theta-1}} + \left(\frac{\alpha I_{2}}{p_{2}(h)}\right)^{\theta} \frac{1}{C_{2,H}^{\theta-1}}, \qquad y^{2}(h) = 0, \quad h \in [0, h_{2}] \\ y^{1}(h) &= \left(\frac{\alpha I_{1}}{p_{1}(h)}\right)^{\theta} \frac{1}{C_{1,H}^{\theta-1}}, \quad y^{2}(h) = \left(\frac{\alpha I_{2}}{p_{2}(h)}\right)^{\theta} \frac{1}{C_{2,H}^{\theta-1}}, \quad h \in (h_{2}, h_{1}) \\ y^{1}(h) &= 0, \qquad y^{2}(h) = \left(\frac{\alpha I_{1}}{p_{1}(h)}\right)^{\theta} \frac{1}{C_{1,H}^{\theta-1}} + \left(\frac{\alpha I_{2}}{p_{2}(h)}\right)^{\theta} \frac{1}{C_{2,H}^{\theta-1}}, \quad h \in [h_{1}, 1] \end{split}$$

Analogous expressions can be obtained for the sector producing low-differentiation goods, a.

Appendix 3. Equations summarizing the solution of the model in the steady state

$$\begin{split} h_1 &= \frac{w_2 \mathcal{E}(1+\tau_{2,h})(\gamma+\lambda) + \delta w_1 - \gamma w_1}{w_1 \lambda + w_2 \mathcal{E}(1+\tau_{2,h})\lambda}, \quad h_2 &= \frac{w_2 \mathcal{E}(\gamma+\lambda) - \delta \mathcal{E} w_2 - \gamma w_{1,h}(1+\tau_{1,h}) + w_2 \mathcal{E}\lambda}{w_1 \lambda(1+\tau_{1,h}) + w_2 \mathcal{E}\lambda}, \\ a_1 &= \frac{w_2 \mathcal{E}(1+\tau_{2,a})(\gamma+\lambda) + \delta w_1 - \gamma w_1}{w_1 \lambda + w_2 \mathcal{E}(1+\tau_{2,a})\lambda}, \quad a_2 &= \frac{w_2 \mathcal{E}(\gamma+\lambda) - \delta \mathcal{E} w_2 - \gamma w_1(1+\tau_{1,a})}{w_1 \lambda(1+\tau_{1,a}) + w_2 \mathcal{E}\lambda}, \\ \mathcal{E} &= \frac{P_1 C_1}{P_2 C_2}, \quad \frac{w_1}{w_2} = 1, \\ M_1 &= x \frac{1}{\beta} P_1 C_1, \quad M_2 = x \frac{1}{\beta} P_2 C_2, \\ P_1 C_1 &\leq \int_0^{h_1} \frac{l_{1,H}}{\gamma + \lambda h} p_1(h) dh + \int_0^{h_2} \frac{l_{1,H}}{\gamma + \lambda h} p_2(h) dh + \int_0^{a_1} \frac{l_{1,A}}{\gamma + \lambda h} p_1(a) da \\ &+ \int_0^{a_2} \frac{l_{1,A}}{\gamma + \lambda a} p_2(a) da + \int_{a_1}^1 \tau_{1,a} \tilde{p}_1(a) \frac{l_{1,A}}{\gamma + \lambda(1-a)} da + \int_{h_1}^1 \tau_{1,h} \tilde{p}_1(h) \frac{l_{1,H}}{\gamma + \lambda(1-h)} dh, \\ P_2 C_2 &\leq \int_{h_2}^1 \frac{l_{2,H}}{\gamma + \lambda(1-h)} p_2(h) dh + \int_{h_1}^1 \frac{l_{2,H}}{\gamma + \lambda(1-h)} p_1(h) dh + \int_{a_2}^{h_2} \frac{l_{2,A}}{\gamma + \lambda(1-a)} p_2(a) da \\ &+ \int_{a_2}^1 \frac{l_{2,A}}{\gamma + \lambda(1-a)} p_1(a) da + \int_0^{a_2} \tau_{2,a} \tilde{p}_2(a) \frac{l_{1,A}}{\gamma + \lambda(1-h)} da + \int_{a_1}^1 \frac{l_{2,A}}{\gamma + \lambda(1-h)} da = (1-\alpha) C_1, \\ \int_{h_2}^1 \frac{l_{2,H}}{\gamma + \lambda(1-h)} dh + \int_{0}^{h_2} \frac{l_{1,H}}{\gamma + \lambda(1-h)} dh = \alpha C_2, \int_{a_2}^1 \frac{l_{2,A}}{\gamma + \lambda(1-a)} da + \int_{0}^{a_2} \frac{l_{1,A}}{\gamma + \lambda(1-h)} da = (1-\alpha) C_2, \\ l_{1,N} &= \delta \left(\int_{h_1}^1 \frac{l_{2,H}}{\gamma + \lambda(1-h)} dh + \int_{a_1}^1 \frac{l_{2,A}}{\gamma + \lambda(1-h)} da \right), l_{2,N} = \delta \left(\int_{0}^{h_2} \frac{l_{1,H}}{\gamma + \lambda h} dh + \int_{0}^{a_2} \frac{l_{1,A}}{\gamma + \lambda h} da \right). \end{split}$$

Appendix 4. Calibrating the parameters γ and λ

Calibrating the parameters γ and λ is a difficult and challenging task since there is no microeconomic evidence to support these values. Further, these values correspond to the technology parameters, and while γ is only a scaling parameter, λ controls the slope of the productivity across products. This means that λ controls the number of nontraded varieties. It is logically easier to calibrate these parameters for a two-country relationship rather than for a crosscountry relationship. However, there is no reason why the domestic technology should be different depending on the trade partner. Since I am calibrating the model using the USA as a benchmark, an important trade partner of this country is needed. Consequently, I chose to calibrate these parameters for the USA-Mexico relationship since both the countries are among each other's biggest trading partners. The model is calibrated for the year 1993, which is the year before the NAFTA took effect. Consider the USA as country 1 and Mexico as country 2. Sector *h* represents the manufacturing sector. The values for γ and λ are obtained as follows:

1. Consider

$$h_{1} = \frac{w_{2,h}\mathcal{E}(1+\tau_{2,h})(\gamma+\lambda) + \delta p_{1,n} - \gamma w_{1,h}}{w_{1,h}\lambda + w_{2,h}\mathcal{E}(1+\tau_{2,h})\lambda},$$

$$h_{2} = \frac{w_{2,h}\mathcal{E}(\gamma+\lambda) - \delta \mathcal{E} p_{2,n} - \gamma w_{1,h}(1+\tau_{1,h})}{w_{1,h}\lambda(1+\tau_{1,h}) + w_{2,h}\mathcal{E}\lambda}.$$
(5)

2. Since $p_1^1(h_1) = \frac{\theta}{\theta-1} w_{1,h}(\gamma + \lambda h_1)$, $w_{1,h}(\gamma + \lambda h_1) = MgC_1(h_1)$, where $MgC_1(h_1)$ represents the marginal cost in country 1 of variety h_1 when the distribution sector is not involved. Hence, in country 2, $w_{2,h}(\gamma + \lambda(1 - h_2)) = MgC_2(h_2)$. These two equations can be rewritten as

$$h_1 = \frac{\frac{MgC_1(h_1)}{w_{1,h}} - \gamma}{\lambda},\tag{7}$$

$$h_2 = 1 - \frac{\frac{MgC_2(h_2)}{w_{2,h}} - \gamma}{\lambda}$$

3. I substitute (5) and (6) in (7) and (8), respectively. Since γ , $\lambda > 0$, the resulting expressions can be written as

$$a(w_{1,h} + w_{2,h}\mathcal{E}(1+\tau_{2,h})) - \delta w_{1,n} = \gamma [2w_{2,h}\mathcal{E}(1+\tau_{2,h})] + \lambda [w_{2,h}\mathcal{E}(1+\tau_{2,h})],$$

$$b(w_{2,h}\mathcal{E} + w_{1,h}(1 + \tau_{1,h})) - \delta w_{2,n}\mathcal{E} = \gamma[2w_{1,h}(1 + \tau_{1,h})] + \lambda[w_{1,h}(1 + \tau_{1,h})],$$

where

$$a = \frac{MgC_1(h_1)}{w_{1,h}}$$
 and $b = \frac{MgC_2(h_2)}{w_{2,h}}$

4. These two equations can be written in the matrix form and solved using a Cholesky decomposition

$$\begin{bmatrix} w_{2,h}\mathcal{E}(1+\tau_{2,h}) & 2w_{2,h}\mathcal{E}(1+\tau_{2,h}) \\ w_{1,h}(1+\tau_{1,h}) & 2w_{1,h}(1+\tau_{1,h}) \end{bmatrix} \begin{bmatrix} \lambda \\ \gamma \end{bmatrix} = \begin{bmatrix} a(w_{1,h}+w_{2,h}\mathcal{E}(1+\tau_{2,h})) - \delta w_{1,n} \\ b(w_{2,h}\mathcal{E}+w_{1,h}(1+\tau_{1,h})) - \delta w_{2,n}\mathcal{E} \end{bmatrix}$$

5. The data required to solve the above operations is obtained as follows:

The average wages per hour in the manufacturing sector for each country are obtained in the World Development Indicators 2005, World Bank.

$$w_{1,h} = w_{USA,1993,h} = 11.74$$
 USD

 $w_{2,h} = w_{MEX,1993,h} = 2.10$ USD

6. The marginal costs are not observable; however, there is a variable that is usually used as a proxy, the unit labor cost (ULC). This measure is an index and gives a relative measure of marginal cost between the USA and Mexico. The ULC for the USA is $ULC_{USA,1994} = 100$ and that for Mexico is $ULC_{MEX,1994} = 15$. The marginal cost in the USA for the variety h_1 , $MgC_1(h_1)$, is not observable. Consequently, I choose $MgC_1(h_1)$ such that the proportion of non-wage costs to total compensation costs is 21%, as in the 1990s (source: International Labour Organization). As such, $MgC_1(h_1) = w_{USA,1993,h} * 1.21 = 14.21$ USD. Due to the differences in the ULCs across countries, $MgC_2(h_2) = 14.21 * 0.15 = 2.13$ USD.

7. The average tariffs between Mexico and the USA before the NAFTA were $\tau_{1,h} = \tau_{US,h} = 2.07\%$, $\tau_{2,h} = \tau_{MEX,h} = 12\%$, and $\delta = 1.403$, such that the distribution margin is between 40% and 80% of the retail price.

8. By using all these data and solving the two equations, I obtain $\gamma = 0.076$ and $\lambda = 2.110$. The restriction $\lambda > \gamma \tau_{1,\alpha} + \delta$ must hold to ensure trade across the countries in both sectors.

Working Papers

00/01 Fernando C. Ballabriga y Sonsoles Castillo: BBVA-ARIES: un modelo de predicción y simulación para la economía de la UEM.

00/02 **Rafael Doménech, María Teresa Ledo and David Taguas:** Some new results on interest rate rules in EMU and in the US.

00/03 Carmen Hernansanz and Miguel Sebastián: The Spanish Banks' strategy in Latin America.

01/01 José Félix Izquierdo, Angel Melguizo y David Taguas: Imposición y Precios de Consumo.

01/02 **Rafael Doménech, María Teresa Ledo and David Taguas:** A Small Forward-Looking Macroeconomic Model for EMU.

02/01 **Jorge Blázquez y Miguel Sebastián:** ¿Quién asume el coste en la crisis de deuda externa? El papel de la Inversión Extranjera Directa (IED).

03/01 Jorge Blázquez y Javier Santiso: México, ¿un ex-emergente?

04/01 Angel Melguizo y David Taguas: La ampliación europea al Este, mucho más que economía.

04/02 Manuel Balmaseda: L'Espagne, ni miracle ni mirage.

05/01 Alicia García-Herrero: Emerging Countries' Sovereign Risk: Balance Sheets, Contagion and Risk Aversion.

05/02 Alicia García-Herrero and María Soledad Martínez Pería: The mix of International bank's foreign claims: Determinants and implications.

05/03 Alicia García Herrero and Lucía Cuadro-Sáez: Finance for Growth: Does a Balanced Financial Structure Matter?

05/04 Rodrigo Falbo y Ernesto Gaba: Un estudio econométrico sobre el tipo de cambio en Argentina.

05/05 **Manuel Balmaseda, Ángel Melguizo y David Taguas:** Las reformas necesarias en el sistema de pensiones contributivas en España.

06/01 **Ociel Hernández Zamudio:** Transmisión de choques macroeconómicos: modelo de pequeña escala con expectativas racionales para la economía mexicana.

06/02 Alicia García-Herrero and Daniel Navia Simón: Why Banks go to Emerging Countries and What is the Impact for the Home Economy?

07/01 **Pedro Álvarez-Lois and Galo Nuño-Barrau:** The Role of Fundamentals in the Price of Housing: Theory and Evidence.

07/02 Alicia García-Herrero, Nathalie Aminian, K.C.Fung and Chelsea C. Lin: The Political Economy of Exchange Rates: The Case of the Japanese Yen.

07/03 **Ociel Hernández y Cecilia Posadas:** Determinantes y características de los ciclos económicos en México y estimación del PIB potencial.

07/04 Cristina Fernández y Juan Ramón García: Perspectivas del empleo ante el cambio de ciclo: un análisis de flujos.

08/01 Alicia García-Herrero and Juan M. Ruiz: Do trade and financial linkages foster business cycle synchronization in a small economy?

08/02 Alicia García-Herrero and Eli M. Remolona: Managing expectations by words and deeds: Monetary policy in Asia and the Pacific.

08/03 José Luis Escrivá, Alicia García-Herrero, Galo Nuño and Joaquin Vial: After Bretton Woods II.

08/04 Alicia García-Herrero and Daniel Santabárbara: Is the Chinese banking system benefiting from foreign investors?

08/05 **Joaquin Vial and Angel Melguizo:** Moving from Pay as You Go to Privately Manager Individual Pension Accounts: What have we learned after 25 years of the Chilean Pension Reform?

08/06 Alicia García-Herrero and Santiago Fernández de Lis: The Housing Boom and Bust in Spain: Impact of the Securitization Model and Dynamic Provisioning.

08/07 **Ociel Hernández y Javier Amador:** La tasa natural en México: un parámetro importante para la estrategia de política monetaria.

08/08 **Patricia Álvarez-Plata and Alicia García-Herrero:** To Dollarize or De-dollarize: Consequences for Monetary Policy.

09/01 K.C. Fung, Alicia García-Herrero and Alan Siu: Production Sharing in Latin America and East Asia.

09/02 Alicia García-Herrero, Jacob Gyntelberg and Andrea Tesei: The Asian crisis: what did local stock markets expect?

09/03 Alicia García-Herrero and Santiago Fernández de Lis: The Spanish Approach: Dynamic Provisioning and other Tools.

09/04 **Tatiana Alonso:** Potencial futuro de la oferta mundial de petróleo: un análisis de las principales fuentes de incertidumbre.

09/05 Tatiana Alonso: Main sources of uncertainty in formulating potential growth scenarios for oil supply.

09/06 Ángel de la Fuente y Rafael Doménech: Convergencia real y envejecimiento: retos y propuestas.

09/07 **KC FUNG, Alicia García-Herrero and Alan Siu:** Developing Countries and the World Trade Organization: A Foreign Influence Approach.

09/08 Alicia García-Herrero, Philip Woolbridge and Doo Yong Yang: Why don't Asians invest in Asia? The determinants of cross-border portfolio holdings.

09/09 Alicia García-Herrero, Sergio Gavilá and Daniel Santabárbara: What explains the low profitability of Chinese Banks?

09/10 **J.E. Boscá, R. Doménech and J. Ferri:** Tax Reforms and Labour-market Performance: An Evaluation for Spain using REMS.

09/11 **R. Doménech and Angel Melguizo:** Projecting Pension Expenditures in Spain: On Uncertainty, Communication and Transparency.

09/12 J.E. Boscá, R. Doménech and J. Ferri: Search, Nash Bargaining and Rule of Thumb Consumers.

09/13 **Angel Melguizo, Angel Muñoz, David Tuesta y Joaquín Vial:** Reforma de las pensiones y política fiscal: algunas lecciones de Chile.

09/14 **Máximo Camacho:** MICA-BBVA: A factor model of economic and financial indicators for short-term GDP forecasting.

09/15 **Angel Melguizo, Angel Muñoz, David Tuesta and Joaquín Vial:** Pension reform and fiscal policy: some lessons from Chile.

09/16 Alicia García-Herrero and Tuuli Koivu: China's Exchange Rate Policy and Asian Trade.

09/17 Alicia García-Herrero, K.C. Fung and Francis Ng: Foreign Direct Investment in Cross-Border Infrastructure Projects.

09/18 Alicia García Herrero y Daniel Santabárbara García: Una valoración de la reforma del sistema bancario de China.

09/19 **C. Fung, Alicia García-Herrero and Alan Siu:** A Comparative Empirical Examination of Outward Direct Investment from Four Asian Economies: China, Japan, Republic of Korea and Taiwan.

09/20 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazábal, Ivonne Ordóñez, Carolina Romero y David Tuesta: Un balance de la inversión de los fondos de pensiones en infraestructura: la experiencia en Latinoamérica.

09/21 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazábal, Ivonne Ordóñez, Carolina Romero y David Tuesta: Proyecciones del impacto de los fondos de pensiones en la inversión en infraestructura y el crecimiento en Latinoamérica.

10/01 **Carlos Herrera:** Rentabilidad de largo plazo y tasas de reemplazo en el Sistema de Pensiones de México.

10/02 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazabal, Ivonne Ordóñez, Carolina Romero, David Tuesta and Alfonso Ugarte: Projections of the Impact of Pension Funds on Investment in Infrastructure and Growth in Latin America.

10/03 Javier Alonso, Jasmina Bjeletic, Carlos Herrera, Soledad Hormazabal, Ivonne Ordóñez, Carolina Romero, David Tuesta and Alfonso Ugarte: A balance of Pension Fund Infrastructure Investments: The Experience in Latin America.

10/04 **Mónica Correa-López y Ana Cristina Mingorance-Arnáiz:** Demografía, Mercado de Trabajo y Tecnología: el Patrón de Crecimiento de Cataluña, 1978-2018.

10/05 **Soledad Hormazabal D.:** Gobierno Corporativo y Administradoras de Fondos de Pensiones (AFP). El caso chileno.

10/06 **Soledad Hormazabal D.:** Corporate Governance and Pension Fund Administrators: The Chilean Case.

10/07 **Rafael Doménech y Juan Ramón García:** ¿Cómo Conseguir que Crezcan la Productividad y el Empleo, y Disminuya el Desequilibrio Exterior?

10/08 **Markus Brückner and Antonio Ciccone:** International Commodity Prices, Growth, and the Outbreak of Civil War in Sub-Saharan Africa.

10/09 Antonio Ciccone and Marek Jarocinski: Determinants of Economic Growth: Will Data Tell?

10/10 Antonio Ciccone and Markus Brückner: Rain and the Democratic Window of Opportunity.

10/11 Eduardo Fuentes: Incentivando la cotización voluntaria de los trabajadores independientes a los fondos de pensiones: una aproximación a partir del caso de Chile.

10/12 Eduardo Fuentes: Creating incentives for voluntary contributions to pension funds by independent workers: A primer based on the case of Chile.

10/13 **J. Andrés, J.E. Boscá, R. Doménech and J. Ferri:** Job Creation in Spain: Productivity Growth, Labour Market Reforms or both.

10/14 Alicia García-Herrero: Dynamic Provisioning: Some lessons from existing experiences.

10/15 Arnoldo López Marmolejo and Fabrizio López-Gallo Dey: Public and Private Liquidity Providers.

10/16 Soledad Zignago: Determinantes del comercio internacional en tiempos de crisis.

10/17 **Angel de la Fuente and José Emilio Boscá:** EU cohesion aid to Spain: a data set Part I: 2000-06 planning period.

10/18 Angel de la Fuente: Infrastructures and productivity: an updated survey.

10/19 Jasmina Bjeletic, Carlos Herrera, David Tuesta y Javier Alonso: Simulaciones de rentabilidades en la industria de pensiones privadas en el Perú.

10/20 Jasmina Bjeletic, Carlos Herrera, David Tuesta and Javier Alonso: Return Simulations in the Private Pensions Industry in Peru.

10/21 **Máximo Camacho and Rafael Doménech:** MICA-BBVA: A Factor Model of Economic and Financial Indicators for Short-term GDP Forecasting.

10/22 Enestor Dos Santos and Soledad Zignago: The impact of the emergence of China on Brazilian international trade.

10/23 **Javier Alonso, Jasmina Bjeletic y David Tuesta:** Elementos que justifican una comisión por saldo administrado en la industria de pensiones privadas en el Perú.

10/24 **Javier Alonso, Jasmina Bjeletic y David Tuesta:** Reasons to justify fees on assets in the Peruvian private pension sector.

10/25 **Mónica Correa-López, Agustín García Serrador and Cristina Mingorance-Arnáiz:** Product Market Competition and Inflation Dynamics: Evidence from a Panel of OECD Countries.

10/26 Carlos A. Herrera: Long-term returns and replacement rates in Mexico's pension system.

10/27 Soledad Hormazábal: Multifondos en el Sistema de Pensiones en Chile.

10/28 **Soledad Hormazábal:** Multi-funds in the Chilean Pension System.

10/29 **Javier Alonso, Carlos Herrera, María Claudia Llanes y David Tuesta:** Simulations of longterm returns and replacement rates in the Colombian pension system.

10/30 Javier Alonso, Carlos Herrera, María Claudia Llanes y David Tuesta: Simulaciones de

rentabilidades de largo plazo y tasas de reemplazo en el sistema de pensiones de Colombia.

11/01 Alicia García Herrero: Hong Kong as international banking center: present and future.

11/02 Arnoldo López-Marmolejo: Forthcoming, Review of International Economics.

The analysis, opinions, and conclusions included in this document are the property of the author of the report and are not necessarily property of the BBVA Group.

BBVA Research's publications can be viewed on the following website: http://www.bbvaresearch.com

Contact details

BBVA Research Mexico Avda. Universidad 1200 Colonia Xoco C.P. 03339 México D.F. Phone: + 52 55 56216620 E-mail: researchmexico@bbva.bancomer.com